COLLECTIVE AGREEMENT

BETWEEN

THE CORPORATION OF THE CITY OF LONDON

and

LOCAL UNION NO. 101 (Canadian Union of Public Employees) London, Ontario

January 1, 2019 - December 31, 2022





Table of Contents

| ARTICLE | TITLE PA | GE NUMBER |
|-----------|--|-----------|
| | 1 - RECOGNITION AND SCOPE | |
| | 2 - UNION SECURITY AND CHECK-OFF | |
| | 3 - MANAGEMENT FUNCTIONS | |
| | 4 - UNION-MANAGEMENT RESPONSIBILITIES | |
| | 5 - UNION REPRESENTATION | |
| | 6 - SENIORITY | |
| | 7 - RECLASSIFICATION AND REVISION OF POSITIONS AND S | |
| | 8 - STAFF CHANGES, ADDITIONS AND PROMOTION | |
| | 9 - REMUNERATION AND INTERVIEW | |
| | 0 - HOURS OF WORK, SHIFT, OVERTIME, CALL OUT, STAND | |
| | MEAL ALLOWANCE | 13 |
| ARTICLE 1 | 1 - HOLIDAYS | 17 |
| | 2 - VACATIONS | 18 |
| ARTICLE 1 | 3 - HOSPITAL, MEDICAL, SICK LEAVE, PENSIONS, | |
| GROUP IN | SURANCE | 21 |
| | 13.1 HEALTH AND WAGE LOSS INSURANCE PLANS | 21 |
| | 13.2 SICK LEAVE | |
| | 13.3 PREVENTATIVE MEDICINE | |
| | 13.4 PENSIONS | |
| | 13.5 LIFE AND WAGE LOSS INSURANCE PLANS | |
| ARTICLE 1 | 4 - REHABILITATIVE ASSIGNMENTS AND WORKPLACE SAF | |
| ADTIOLE 4 | AND INSURANCE | 28 |
| | 5 - GRIEVANCE PROCEDURE | |
| | 7 - DISCHARGE AND DISCIPLINE CASES | |
| | 8 - PROTECTION OF EMPLOYEE'S POSITION | |
| | 9 - CAR AND CLOTHING ALLOWANCE | |
| | 0 - UNION CONFERENCE, CONVENTION OR SEMINAR | |
| | 1 - LEAVES OF ABSENCE | |
| | 2 - BEREAVEMENT LEAVE | |
| | 3 - TEMPORARY ASSIGNMENT | |
| | 4 - PART TIME EMPLOYMENT | |
| ARTICLE 2 | 5 - STUDENT EMPLOYMENT | 39 |
| | 6 - NO STRIKES OR LOCK-OUTS | |
| | 7 - EDUCATION ALLOWANCE | |
| | 8 - LAY-OFF AND RECALL | |
| | 9 - GENERAL | |
| | 0 - TERMS AND DEFINITIONS | |
| | 1 - TERM OF AGREEMENT | |
| | E A - POSITION CLASSIFICATION AND SALARY SCHEDULE | |
| | E B - CLOTHING AND SAFETY EQUIPMENT | |
| | A - RETIREES BENEFITS | |
| APPENDIX | B - LONG TERM DISABILITY DETAILS | /0 |

| LETTERS OF UNDERSTANDING | |
|--|------------------|
| Leaves of Absence | |
| Ontario Works Program | |
| Reduced Hours of Operation (Holiday Closure) | |
| Testing | |
| Flexible Work Schedule Program and Compensatory Time Guidelines | 79 |
| Convention Servicing | |
| Terms of Dedicated Presidency for Local 101 | 84 |
| Municipal Law Enforcement Officer II | 86 |
| Job Sharing | 87 |
| Call In & Stand-by – Remote Access | 89 |
| Excess Hours of Work Agreement | 90 |
| United Way Sponsored Employee Program | 92 |
| High School Co-op Placement | 94 |
| Seasonal Tourism Counsellor/Paid Lunch Under Article 10 | 95 |
| Referral to Arbitration of a Job Evaluation Matter | 96 |
| Employees hired on or before July 22, 2015 – Job Posting and "Bumping" | 98 |
| FORMER PUC – APPENDIX AGREEMENT "1" | 100 |
| ARTICLE 13 - HOURS OF WORK | |
| ARTICLE 14 - PREMIUM PAY | |
| ARTICLE 23 - SICK AND ACCIDENT PLAN | |
| ARTICLE 24 - MEDICAL PLANS | |
| ARTICLE 25 - WORKPLACE SAFETY AND INSURANCE BENEFITS | |
| ARTICLE 26 - PENSION AND INSURANCE PLANS | |
| ARTICLE 27- EMPLOYMENT INSURANCE PREMIUM | |
| | |
| APPENDIX SCHEDI II E "A" | |
| APPENDIX SCHEDULE "A" | |
| FORMER TOWN OF WESTMINSTER EMPLOYEES BENEFITS – APPENDIX | 107 |
| FORMER TOWN OF WESTMINSTER EMPLOYEES BENEFITS – APPENDIX AGREEMENT "2" | 107 |
| FORMER TOWN OF WESTMINSTER EMPLOYEES BENEFITS – APPENDIX | 107 |
| FORMER TOWN OF WESTMINSTER EMPLOYEES BENEFITS – APPENDIX AGREEMENT "2" BENEFITS AT A GLANCE FORMER MIDDLESEX COUNTY EMPLOYEES BENEFITS - APPENDIX AGRI | 107108109 EEMENT |
| FORMER TOWN OF WESTMINSTER EMPLOYEES BENEFITS – APPENDIX AGREEMENT "2" BENEFITS AT A GLANCE | 107108109 EEMENT |

THIS AGREEMENT RATIFIED ON THE 16 day of October, 2018.

BETWEEN:

THE CORPORATION OF THE CITY OF LONDON (hereinafter called the Corporation)

OF THE FIRST PART

-and -

CANADIAN UNION OF PUBLIC EMPLOYEES, LOCAL UNION NO.101 (hereinafter called the Union)

OF THE SECOND PART

Whereas in the interest of the efficient conduct and administration of the City's affairs, it is desirable and necessary that there shall obtain harmonious relations between the City Council, the Managing Directors of the Departments and City employees; fair and reasonable remuneration for the services rendered, having regard to the responsibility attached to the position held, the nature of the duties thereof, the manner of their discharge and seniority in the service, security of tenure of office and promotion within the service. This Agreement is entered into by the parties hereto in order to provide for orderly collective bargaining relations between the Corporation and its employees. It is the desire of both parties to co-operate in maintaining a satisfactory relationship between the Corporation and its employees, and to provide an amicable method of settling any difference or grievance relating to the general working conditions which may arise from time to time.

NOW THEREFORE to effectuate the foregoing, the Corporation hereby covenants and agrees with the Union as follows:

ARTICLE 1 - RECOGNITION AND SCOPE

- 1.1 The Corporation recognizes the Union as the exclusive bargaining agent for all the Corporation's employees save and except the following:
 - Those employees of the Corporation who are represented as bargaining agent by Local No. 107 (Outside Workers), or by Local No. 220, Service Employees International Union, Dearness Home, Unifor, Local 302 or by The London Professional Firefighters' Association.
 - Dr. John Dearness Home Management and Administrative staff.

- And, with respect to positions within the former Parks and Recreation Department, staff at the Golf Courses, program staff at the Aquatic Centre (except full time Deck Supervisors) and casual staff defined as employees working less than twenty four (24) hours per week and employees hired for the period between April 1 and Thanksgiving who perform work other than that evaluated and classified in Schedule "A".
- Security Guards.
- Those Management and Administrative positions exempted by virtue of the provisions of the <u>Ontario Labour Relations Act</u> or agreement of the Parties, a list including the name of the person occupying the position, which shall be provided to the Union by December 31st each year.

ARTICLE 2 - UNION SECURITY AND CHECK-OFF

- 2.1 All present employees shall become or remain, as the case may be, members of the Union; and all persons who may hereafter become employees covered by this Agreement, shall become members after ninety (90) calendar days of employment and shall maintain such membership, all as a condition of continuing employment.
- 2.2 Once each month so long as this Agreement continues to operate, the Corporation will deduct from the remuneration of each employee who is covered by this Agreement and to whom any remuneration is due in that month, an amount equal to the employee's regular monthly Union dues or part thereof. The Union shall notify the Director of People Services of the Corporation in writing of the amount of such dues or such part thereof from time to time.
- 2.3 In addition to the dues in Article 2.2, the Corporation shall similarly deduct from the remuneration of each such employee such sum as may constitute the total of any monthly assessments adopted by the Union as a contribution to the social and general welfare of the Union. The Union shall notify the Director of People Services of the Corporation in writing of the amount of any such assessments.
- 2.4 All sums deducted pursuant to this Article shall be remitted by the Corporation to the Treasurer of the Union once each month together with a list of names of all employees from whose remuneration Union dues and assessments were so deducted. The list of names shall also indicate the amount of Union dues deducted for each employee, employees' annual salary in their base classification for full time employees, and employees' bi-weekly earnings for the applicable month for part time employees. The Corporation shall notify the Union of terminations of employment and of newly hired employees in the pay period following the pay period in which the status of employment changed.
- 2.5 The Union shall indemnify and save the Corporation harmless from and against all claims and demands brought or made against the Corporation by an employee as a result of the deductions and remittance by the Corporation to the Union of dues and assessments pursuant to this Article; provided that this section does not apply to a

request by the Union for correction and adjustment of any error in the deduction or remittance of Union dues or assessments.

- 2.6 The Corporation will inform affected employees of the provisions in this article and give all permanent employees and all temporary employees hired for more than twelve (12) weeks a copy of the Collective Agreement. During the first day at work the employees noted above will be introduced by a Manager to the Union Steward and the Worker Health and Safety Representative, and shall be allowed fifteen (15) minutes to meet with each
- 2.7 When employees who were paying Union dues prior to being laid off are returned to work within twelve (12) months of being laid off, the employees will start paying union dues upon return to work in accordance with Articles 2.2 and 2.3.
- 2.8 The Union shall have the right to post notices and bulletins that may be of interest to the Union and its members on Corporate Bulletin Boards and on the Corporate Intranet, subject to the approval of the Director of People Services or designate. Permission is not required for routine postings of meetings, seminars and conferences.
- 2.9 Employees shall be permitted to solicit membership for the Union on Corporation property outside of actual hours of assignment of any employee soliciting or being solicited and during recognized breaks.

ARTICLE 3 - MANAGEMENT FUNCTIONS

- 3.1 The Union recognizes the rights conferred upon the Corporation by Statute and the rights of the Corporation to hire, lay-off, promote, demote, transfer, suspend or otherwise discipline and discharge an employee for proper cause; provided that a claim of discriminatory promotion or demotion, or a claim that an employee has been discharged or disciplined without proper cause, may be the subject of a grievance and dealt with under the provisions of Article 15 of this Agreement.
- 3.2 The Union further recognizes the undisputed right of the Corporation to operate and manage its business in all aspects in accordance with its responsibilities. In addition, the location of its plants or places of employment, the methods, processes and means of performing the various works are solely and exclusively the right and responsibility of the Corporation. The Corporation also has the right, and the Union recognizes it, to make and alter, from time to time, the rules and regulations to be observed by the employees, which rules and regulations shall not be contrary to the provisions of this Agreement.
- 3.3 The Corporation recognizes the foregoing Articles 3.1 and 3.2 are subject to such provisions, regulations and/or restrictions governing the exercise of these rights as are provided in this Agreement and are subject to the right of the Union and/or the employee concerned to lodge a grievance in the manner and to the extent herein provided.
- 3.4 The Corporation endorses the right of every employee to work in an environment free from harassment and employees are free to pursue all avenues in the Corporate

policy and the Collective Agreement, including the grievance procedure for resolving complaints of harassment that may arise.

ARTICLE 4 - UNION-MANAGEMENT RESPONSIBILITIES

- 4.1 All employees agree to give their best efforts at all times to the performance of their work and will not in any circumstances deliberately delay, shirk or cause delay to any work through grievances but will carry on with their work while any grievance is being investigated. Managing Directors will not discriminate against employees who have requested investigation into an alleged grievance and all parties hereto will at all times extend the fullest co-operation to one another in order that the assigned work shall be carried on economically.
- 4.2 The Corporation and the Union agree that there shall be no discrimination, interference, restriction, or coercion exercised or practised with respect to any employee in the matters of hiring, wage rates, training, up-grading, promotion, transfer, lay-off, recall, discipline, discharge or in the administration of any of the provisions of this Collective Agreement by reason of any grounds prohibited under the <u>Ontario Human Rights Code</u>, nor by reason of the employee's membership or activity in the Union, or for any other reason.
- 4.3 The Corporation and the Union recognize that this Agreement is subject to all government regulations as they apply to employees who have served, are now serving, or may in the future serve in the Armed Forces.
- 4.4 The Corporation and the Union acknowledge that the <u>Ontario Human Rights Code</u>, the <u>Employment Standards Act</u>, the <u>Ontario Labour Relations Act</u>, and the <u>Occupational Health and Safety Act</u> shall apply to all employees. Any greater right or benefit contained in these Acts shall prevail.

ARTICLE 5 - UNION REPRESENTATION

- 5.1 Subject to Article 5.2 of this Agreement, the Corporation will recognize the following Committees of employees for the respective purposes shown. All Committees will consist of equal representation of the Union and the Corporation. Subject to Article 5.3 of this Agreement, employees shall not suffer any loss of remuneration in performing any committee work associated with the Committees listed as follows:
 - (a) The Bargaining Committee shall consist of not more than six (6) representatives of the Union for the purpose of negotiating this Agreement and its renewal
 - (b) The Union Management Committee shall consist of three (3) Union and three (3) Management representatives. At least two of the active Management members will be Human Resources Representatives and at least two of the active Union members will be members of the Union's

Executive. The purpose of this Committee is to improve relations between the Corporation and its employees from the bargaining unit by making recommendations which will create a better working environment and improve services and by making recommendations on conditions causing grievances and misunderstandings. This Committee shall not have jurisdiction over wages or any other matter relating to collective bargaining, including the administration of this Agreement. This Committee shall not supersede the activities of any other Committee of the Union or the Corporation and it shall not have the power to bind either the Union, its members or the Corporation to any decisions or conclusions reached in its discussions. This Committee shall have the power to make recommendations to the Union and to the Corporation with respect to its discussions and conclusions. Necessity for a meeting will be indicated by a letter from one party to the other containing an agenda of the subjects to be discussed. The parties will meet at a mutually acceptable time.

- (c) The Joint Job Evaluation Committee shall consist of not more than six (6) Union and not more than six (6) Management representatives as required for the purpose of measuring the relative value of jobs within the scope of the bargaining unit. This Committee shall evaluate jobs within the employee group.
- (d) The Joint Clothing Committee shall consist of three (3) Union and three (3) Management representatives for the purpose of determining the nature and specifications of clothing and safety equipment outlined in Article 19 and Schedule "B" of this Agreement.
- (e) The Joint Health and Safety Committee City Hall shall consist of four (4) Union and four (4) Management representatives for the purpose of performing the duties prescribed by the <u>Occupational Health and Safety Act</u>, R.S.O. 1990, Chapter 0.1, and as further outlined in the Terms of Reference, jointly agreed to by the parties.
- (f) The Joint Health and Safety Committee Citi Plaza shall consist of six (6) Union and six (6) Management representatives for the purpose of performing the duties prescribed by the <u>Occupational Health and Safety Act</u>, R.S.O. 1990, Chapter 0.1, and as further outlined in the Terms of Reference, jointly agreed to by the parties.
- (g) The Joint Health and Safety Committee Satellite Centres shall consist of four (4) Union and four (4) Management representatives for the purpose of performing the duties prescribed by the <u>Occupational Health and Safety Act</u>, R.S.O. 1990, Chapter 0.1, and as further outlined in the Terms of Reference, jointly agreed to by the parties.
- (h) Any other Joint Health & Safety committee that may be required to comply with the provisions of the <u>Occupational Health and Safety Act</u>, R.S.O. 1990, Chapter 0.1.

- (i) The Grievance Committee shall consist of not more than three (3) Union representatives for the purpose of investigating and processing grievances as outlined in Articles 15 & 16 of this Agreement.
- (j) The Return To Work Committee shall consist of up to four (4) Union and up to four (4) Management representatives for the purpose of returning those employees from the bargaining unit with occupational or non- occupational disabilities or diminished capacity to gainful employment, with the main objective to return those employees to their regular predisability work.
- (k) The Joint Charity Chest Fund Committee shall consist of four (4) Union and four (4) Management representatives for the purpose of encouraging employees from the bargaining unit and Management to make charitable donations, predominately through payroll deductions and special events, and to grant such donations to local charities on behalf of the employees.
- (I) The Classification Review Committee shall consist of four (4) Management representatives and four (4) representatives for the Union for the purpose of reviewing positions excluded from the Bargaining Unit. The Committee shall operate in accordance with the related terms of reference dated January 21, 2003.
- 5.2 (a) The Corporation will not be required to recognize or deal with employees on any of the Committees in Article 5.1 unless those employees have acquired seniority under Article 6 and the Union has notified the Director of People Services in writing of the names of such employees and the Committees of which they are members from time to time. The Union will also notify the Director of People Services in writing on an annual basis, or as changes occur, of the names of the Executive and Stewards of the Union.
 - (b) The Corporation will notify the Union on an annual basis of its representatives on the Committees listed in Article 5.1.
- 5.3 No employee who is a member of the Executive or other Committee listed in Article 5.1 shall leave their work on Union business without first obtaining permission from their appropriate Management supervisor. Committee members shall receive their regular pay for such leaves of absence only if such leave pertains to this Agreement and consists of meeting with Management representatives during normal hours of work. Permission for such leave from the immediate Management supervisor will be granted unless there are extenuating circumstances which would adversely affect the services of the Corporation. In the event of permission not being granted the reasons for refusal will be confirmed in writing by the immediate Management supervisor.
- 5.4 A full time representative of the Union may attend meetings of any Committees of employees.
- 5.5 An employee may be a member of more than one of the said Committees.

- 5.6 It is the policy of the Parties that all correspondence between them, other than grievances or correspondence arising out of grievances, shall pass to and from the Director of People Services and the Secretary of the Union.
- 5.7 All agendas of the City Council are to be sent to the Union at the same time they are sent to the members of the said Council. All minutes, if any, of the meetings of the Council are to be similarly sent to the Union, provided that the Union shall not be entitled to receive any minutes concerning or relating to meetings of the said Council which are held in camera

ARTICLE 6 - SENIORITY

- 6.1 (a) As used in this Agreement, "seniority" means length of continuous service with the Corporation, calculated from the date upon which the employee last commenced employment with the Corporation.
 - (b) Seniority as follows:
 - (i) Where two or more employees have the same effective date of hire the precedence in position on the seniority list will be determined by lottery.
 - (ii) All affected employees must be present, numbers will be drawn and sub-seniority dates established.
 - (iii) The affected employees will sign a letter acknowledging their subseniority date.
- 6.2 An employee shall be considered a probationary employee until the employee has worked sixty-five (65) days (which period may be extended by mutual agreement of the parties in writing) after which the employee's name shall be placed on the seniority list mentioned in Article 6.3 and the employee's seniority shall date back to the date of the employee's last hiring.
- 6.3 The Corporation will maintain a seniority list showing each employee's name, department, the date upon which seniority commenced and job classification. The seniority list will be updated by the Corporation quarterly and posted on the Intranet and on all Bulletin Boards. Complaints about the accuracy of a seniority list will be considered within thirty (30) days of the date of such posting and if no complaint or grievance is received within that time, the list shall then be deemed to be accurate.
- 6.4 Employees who are absent from work due to illness or accident for less than one year (which may be extended for further periods of time) or who are on approved leave of absence or serving in the Armed Forces shall continue to accumulate seniority during the period of such absence or service.

- 6.5 All seniority rights of an employee shall cease for any one of the following reasons:
 - (a) The employee resigns.
 - (b) The employee is discharged and not reinstated through the grievance or arbitration procedures.
 - (c) The employee fails to return from leave of absence without notifying the Corporation at least twenty-four (24) hours prior to the date of the expiry of the leave of absence, provided such notification is reasonably possible.
 - (d) The employee is absent from work without permission for five (5) or more than five (5) consecutive working days unless such absence is proven to the satisfaction of the Corporation to have been due to causes beyond the employee's control.
 - (e) The employee is retired pursuant to the terms of this Agreement or retires.
- 6.6 An employee who accepts a temporary position outside of the bargaining unit shall retain earned seniority ("retained seniority") subject to the following rules:
 - (a) Retained seniority shall be calculated as of the day the employee leaves a union position for a non-union position.
 - (b) No additional seniority shall accumulate until the employee returns to a union position.
 - (c) While the employee is in a non-union position, retained seniority shall not be recognized for any purpose and specifically shall not be used for job posting or layoff purposes.
 - (d) Notwithstanding any other provisions of this Agreement, no employee in a union position shall be displaced as a result of the return of any employee to the bargaining unit (with retained seniority).
 - (e) If the employee with retained seniority is successful under Article 8, the retained seniority will immediately be valid for all purposes.
 - (f) An employee transferred out of the bargaining unit can be returned to their previous permanent position if the return occurs within twenty-four (24) months of transfer.
- 6.7 An employee who accepts a permanent position outside the bargaining unit shall forfeit all rights covered under this Agreement.

ARTICLE 7 - RECLASSIFICATION AND REVISION OF POSITIONS AND NEW POSITIONS

- 7.1 The Parties agree to the provisions of the Job Evaluation Booklet which is considered to form part of the Collective Agreement between the parties.
- 7.2 Prior to finalization of redundancies, a subcommittee of the JE Committee will be established to interview the manager regarding the impact and any redistribution of duties, for the purpose of determining the ramification on jobs.

 Whenever the Corporation proposes to reclassify or revise any of the jobs in Schedule " A " to this Agreement or to establish a job of a nature not already classified in this Agreement; or to declare redundant any of the said jobs and thereby transfer any of the functions of the redundant job to other jobs within the bargaining unit; the foregoing shall be done in accordance with the Job Evaluation Booklet. Employees may request reevaluation of their jobs in accordance with the terms of the Job Evaluation Booklet.
- 7.3 The Corporation shall maintain and keep current a database of job descriptions which have been agreed to in writing by the Parties. The Corporation shall ensure that the Union has access to this database in order that it may obtain copies of revised or new job descriptions. Subject to Article 7, the job descriptions mentioned in this Article do not form part of this Agreement and are intended for the guidance of the Parties.
- 7.4 The Parties agree that the Job Evaluation process is not intended to circumvent the promotion procedures as provided in this Agreement.

ARTICLE 8 - STAFF CHANGES, ADDITIONS AND PROMOTION

- 8.1 (a) Whenever a new job is established in accordance with Article 7, or, there is a permanent vacancy in any of the positions covered by this Agreement and the Corporation proposes to fill such vacancy, the following shall apply:
 - The Corporation will post up a notice of the new position or permanent vacancy in all Departments of the Corporation for a period of five (5) working days excluding the date of posting.
 - In this Agreement the expression "permanent vacancy" means a
 vacancy caused by such events as promotion, resignation, retirement or
 discharge, and which is indefinite or long lasting in nature and does not
 include a vacancy caused by approved or authorized absence from work
 of an employee.
 - (b) The notice will contain the name of the department, a summary of the job description for the position, the number of employees required to fill the position, the salary range or rate of pay, weekly days and hours of work, and the name of the official to whose attention applications are to be directed, and, where the position is under review by the Joint Job Evaluation Committee, a notation indicating that status.

- (c) Employees may apply for a permanent position vacancy in a salary range equal to their present position, or, employees may apply for any permanent position in a salary range which is lower than their own. The requirements of this Article shall apply.
- 8.2 (a) Employees who wish to apply for any posted vacancy shall make application in writing (supplying the information required by the Corporation on forms supplied by the Corporation or on electronic format) to the Director of People Services during the period of five (5) working days mentioned in Article 8.1 (a) and shall set out their qualifications in their applications. Except for permanent employees in the trial period for a non-permanent position, employees in a trial period are ineligible to apply for a posted vacancy unless the posting would result in a promotion to a higher job classification. Any successful applicant will be selected within a reasonable time after such posting as follows:
 - Applicants will be considered on the basis of the following factors: education, training, work experience, ability and appraisal of past performance. In the event that two (2) or more applicants are relatively equal with respect to all of these factors, seniority shall govern.

The Corporation will notify the Union in writing if the most senior applicant will not be appointed to any such vacancy.

If there is no applicant or no successful applicant from the Bargaining Unit, the Corporation may then fill the permanent vacancy from outside the Bargaining Unit.

- (b) Whenever any posted vacancy is filled, a notice containing the name of the employee promoted or the name of the newly-hired employee, the position in question and the effective date of the appointment shall be posted on all bulletin boards within and for five (5) working days and a copy of the notice shall be forwarded to the Secretary of the Union as soon as practicably possible following such appointment; provided the employee shall, subject to Article 8.3, be paid the applicable rate for such position commencing on the date the employee was required to assume the duties thereof.
- (c) If no one is placed, pursuant to Article 8.3, in the permanent vacancy or vacancies, as the case may be, within sixty (60) working days after the date of such posting, both the posting and any application shall be deemed to have been withdrawn.
- 8.3 (a) The successful applicant under Article 8.1 (a) will be placed in the permanent vacancy for a trial period of sixty five (65) days worked. The trial period may be extended by mutual agreement of the Parties in writing and such agreement shall not be unreasonably withheld. If the employee proves satisfactory, the employee will then be confirmed in the employee's new

classification. Management may, in its discretion, confirm such an employee after the employee has worked forty (40) days in the new classification. If the employee proves unsatisfactory during the trial period, or if the employee is unable to perform the new duties, the employee will be returned to the employee's former position at the employee's former salary or rate of pay, as will any other employee in the Bargaining Unit who was promoted or transferred by reason of such placing. The status of temporary employees who are so returned to the employee's former position or duties, shall not be altered as a result of such placing and return, nor shall they thereby acquire seniority, notwithstanding any other provision of this Agreement.

(b) If the successful applicant is returned to the employee's former position under this clause, the Corporation will give consideration to those employees who are unsuccessful applicants for the initial vacancy and should the Corporation place any employee in the permanent vacancy, paragraph (a) of this clause shall apply.

ARTICLE 9 - REMUNERATION AND INTERVIEW

- 9.1 (a) The scale of remuneration set out in Schedule "A" shall apply during the term of this Agreement. No amendment or change in the rates or classifications in Schedule "A" shall become effective until agreed upon by the parties or as implemented pursuant to Article 7.
 - (b) Employees covered by this Agreement shall be paid bi-weekly.
 - (c) Salary increases through the salary progression steps shall be as provided in Schedule "A".
- 9.2 The Corporation, through the respective Managing Director, or an appointee, may interview each employee periodically but no more than three (3) times per year with the appraisal form to be read by the employee. The employee will be provided a copy of the completed appraisal form.
- 9.3 When employees are assigned on a temporary basis for more than five (5) consecutive working days to perform the duties of a job in a higher classification, they will be paid the next higher of the two rates for the replacement period retroactive to the first day of such replacement. However, if the next higher rate is not at least eight hundred dollars (\$800.00), higher than their annual rate, their annual rate shall be adjusted by a minimum of eight hundred dollars (\$800.00). No temporary assignment shall be utilized to deprive employees with seniority of the opportunity of appointment to a position per the provision of Article 6 or Article 8.
- 9.4 (a) When employees are promoted from one job classification to another, they shall be paid the next higher rate as provided in Schedule "A".

However, if the next higher salary rate does not result in at least, eight hundred dollars (\$800.00) increase, their salary shall be adjusted by an amount of a minimum of eight hundred dollars (\$800.00). The adjustment may result in the employee receiving a salary which does not appear in Schedule "A". Future salary increases shall be in accordance with Schedule "A".

- (b) When employees are the successful applicant for positions in lower classifications than their own, or demoted to a position in a classification lower than their own, employees shall be paid the job rate of the lower classification, if employees are at the job rate in the current position. If employees are at an intermediary salary rate, they shall be placed at the salary rate closest to their own in the new classification.
- 9.5 (a) Where employees move from one job classification to a higher classification on a permanent basis, through the provisions of Article 8, the initial rate of pay shall be as determined under Article 9.4 and lateral salary increases shall be adjusted to the next higher rate in the new classification on each anniversary date or semi-annual (whichever is applicable) date of the appointment there to.
 - (b) (i) Where employees move from one job classification to a higher job classification on a permanent basis, through the provisions of Article 7, the rate of pay in the new classification shall be at the same salary progression step as the employee was in the previous classification. As a result of re-evaluation, the employee's anniversary date remains unchanged.
 - (ii) Where employees move from one job classification to a lower job classification by the Joint Job Evaluation process, their salary shall be "red-circled" at their current rate of pay until the salary rate in the new classification reaches and surpasses the "red-circled" rate.
 - (iii) If red-circled employees' positions are declared redundant and they bump into a position with the same classification, they will maintain their red-circled salary.
 - (c) Where permanent employees are hired by the Corporation and receive a rate of pay greater than the starting rate in the job classification, they shall be credited, for the purpose of lateral salary increases, with having completed that period of service in the new classification to which their new rate of pay relates, and lateral salary increases shall accordingly be adjusted to the next higher rate within the classification on the anniversary date or semi-annual (whichever is applicable) date of appointment thereto.
- 9.6 (a) When permanent employees are engaged in a temporary assignment and return to their regular classification, they will return at the rate of pay which

they would have normally received had they continued in their regular classification.

- (b) When permanent employees are engaged in a temporary assignment and are the successful applicants for the same posted temporary or permanent position within a three (3) month period, they will receive the higher of the rate of pay they would have normally received had they continued in the original temporary assignment, or the rate under Article 9.4 (a).
- 9.7 If permanent employees on a temporary assignment are the successful applicant for the same job on a permanent basis under the relevant terms of Article 6 or of Article 8, and such permanent appointment is consecutive to the temporary assignment, the permanent anniversary date of appointment shall be established as the date on which the temporary assignment commenced.
- 9.8 Employees temporarily assigned by Management to a position with a lower pay classification shall be paid at their regular rate.

ARTICLE 10 - HOURS OF WORK, SHIFT, OVERTIME, CALL OUT, STAND-BY, MEAL ALLOWANCE

- 10.1 (a) A normal work week shall be thirty-five (35) hours consisting of any five (5) consecutive seven (7) hour normal work days, Monday to Saturday inclusive. The normal work day will be scheduled on an eight (8) hour basis, inclusive of one (1) unpaid hour off for lunch as follows:
 - (i) Monday through Friday, not to commence before 7:00 a.m. nor finish later than 8:00 p.m.; and
 - (ii) Saturdays not to commence before 8:30 a.m. nor finish later than 4:30 p.m.

Lunch hours may be on a staggered basis as arranged by the Managing Director, or designate, except in cases of emergency, no employee's lunch hour shall commence earlier than after three (3) hours of work or later than after five (5) hours of work.

(b) It is recognized that different work areas within the Corporation may require different hours of work and that the hours of work within the normal work week and normal work day as defined in Article 10.1(a) may vary amongst and within work areas. When assigning normal work days and normal work weeks, the Corporation will take into consideration any expressions of interest by employees in the applicable classification(s) and work area to work the required hours. If the required hours of work are not filled after consideration of expressions of interest, the Corporation will assign the hours of work under Article 10.1(a) to the employees(s) in the classification

and work area by reverse order of seniority (i.e. from the least to the most senior) and/or to temporary employees in the classification.

The Corporation will post a schedule of work days and hours of work under Article 10.1(a) for each classification within each work area at least twice per calendar year. Such schedule(s) will be posted at least forty-five (45) calendar days' in advance of the date the applicable schedule comes into effect.

- (c) Separate hours of work are provided in the appendix for employees transferred from CUPE Local 4. When a former CUPE Local 4 position becomes permanently vacant and the Corporation intends to fill the vacancy, a meeting will be convened between Management and the Union for the purpose of discussing the hours of work of the position to be posted. The hours of work will be governed by the needs of the operation in which the position is situated.
- 10.2 (a) As used in this Article, the expression "shift" shall mean any eight (8) hour period made up of seven (7) working hours and one (1) hour off for lunch, for any consecutive five (5) day period, Monday to Saturday, other than the normal work day as set out in Article 10.1 of this Agreement. A day on which a shift falls shall be determined by the day on which it commences. For the purpose of this Agreement, only the following jobs are to be scheduled on such shifts:
 - Supply Services Representative
 - Inventory Control Clerk
 - Provincial Offences Officers

The Supply Services Representative may be required to work a split shift only under extenuating circumstances and not to exceed seven (7) hours per day. The Supply Services Representative will be paid double (2x) the applicable shift premium when required to work a split shift.

- (b) A "shift" for the Facilities Services Custodial or Maintenance Staff shall mean any seven and one half (7½) hour period made up of seven (7) working hours and one half (½) hour off for lunch, for any consecutive five (5) day period, Monday to Saturday other than the normal work day as set out in Article 10.1 of this Agreement. A day on which a shift falls shall be determined by the day on which it commences.
- (c) A "shift" for the employees in the Information Technology Services Division and for Tourism London Inc., shall mean any eight (8) hour period made up of seven (7) working hours and one (1) hour off for lunch, for any consecutive five (5) day period, other than the normal work day as set out in Article 10.1 of this Agreement. A day on which a shift falls shall be determined by the day on which it commences.

- (d) Notwithstanding the provisions of Article 10.1 (a), Environmental Control Laboratory staff may be required to work shifts to cover a twenty-four (24) hour period for a maximum of ten (10) times a year for the purpose of conducting twenty-four (24) hour surveys. Should such shifts be required, employees will receive the applicable overtime premium as overtime to be banked and observed as time off in accordance with Article 10.8(b).
- (e) Notwithstanding the provisions of Article 10.1 (a), Management may set shifts starting no later than 2:00 pm with double shift premium for employees assigned to attend regularly scheduled meetings of Council, standing committees and subcommittees of Council.
- 10.3 All hours of shift shall be deemed to be included in the calendar day on which the shift started.
- 10.4 (a) An employee who is required to work a shift shall be paid in addition to their normal rate a shift premium of one dollar and twenty cents (\$1.20) for each hour worked after 8:00 p.m. and before 7:00 a.m. Monday through Friday and after 4:30 p.m. and before 8:30 a.m. Saturday, but not for absence from work by reason of illness.
 - (b) In addition to the shift premium referred to in part (a), a weekend premium of one dollar (\$1.00) per hour in addition to the employee's normal rate of pay for all regular hours worked between midnight Friday and midnight Sunday, but not for absence from work by reason of illness.
- 10.5 The expression "overtime" shall mean any period of time worked outside a normal work day or shift, or outside thirty-five (35) hours in any work week (Monday to Saturday), but not both, and shall not include call out under Article 10.9. Where practical, overtime opportunities will be offered on as equitable basis as possible within job classification, work area and work assignment.
- 10.6 (a) Any employee who is required to work overtime or on a holiday, as the case may be, shall be paid as follows:
 - (i) On any of Monday through Saturday time and one-half $(1\frac{1}{2}x)$.
 - (ii) On a Sunday double time (2 x).
 - (iii) On a holiday double time (2 x) in addition to holiday pay.
 - (b) No employee shall receive both overtime payment and shift premium for the same hours worked.
 - (c) No employee shall be laid off work in any week merely for the reason that they have worked overtime.

- 10.7 No employee shall work overtime unless authorized by a Managing Director or delegate.
- 10.8 (a) Managing Directors may, in their discretion, allow time off to employees when they request casual time off for a particular purpose provided the employees agree to make up the time on an hour for hour basis. Such time off shall not be utilized to extend annual vacations or circumvent payment of overtime and shall be limited to a maximum of three (3) hours in any one (1) instance.
 - (b) Instead of a cash payment for overtime, employees may choose to receive time off at the appropriate overtime rate at a time mutually agreed upon between the employees and their Managing Director up to a maximum of five (5) days to be used by calendar year end. The five (5) day maximum may be exceeded with Management approval. In extenuating circumstances if accumulated overtime is unable to be observed, the accumulated amount will be paid out by calendar year end.

Any time off in lieu of cash payment accumulated in the month of December will be paid out by calendar year end unless the employee requests for it to be carried over, in which case the time off in lieu must be used by March 31st of the following year or it will be paid out at that time at the rate at which it was earned.

- 10.9 Employees called to work outside of their normal working hours and/or shift shall be paid a minimum of three (3) hours at the applicable premium rate or the actual time worked at the applicable premium rate, whichever is the greater.
- 10.10 Employees covered by this Agreement who are required to work overtime, as defined in Article 10.5 and Appendix 1, Article 14.01, shall be entitled to a meal allowance of fourteen dollars (\$14.00) provided they work:
 - (a) A minimum of two (2) hours overtime consecutive to the normal work day or shift. or
 - (b) A minimum of three and one half (3½) hours cumulative overtime in one (1) day either immediately following the normal work day or shift or following a two (2) hour interval outside their normal work day or shift.
 - (c) A minimum of three and one half $(3\frac{1}{2})$ hours overtime on a regular day off.
- 10.11 Employees shall be entitled to a fifteen (15) minute break period in each half of each normal work day or shift, as the case may be.
- 10.12 Employees who are on stand-by shall be paid while on stand-by at the regular rate of pay for the following hours:

- two (2) hours for each of Saturday and Sunday.
- four (4) hours for each of the holidays in Article 11.
- one (1) hour for each evening (evening being defined as all of the period of time from the cessation of normal working hours on one (1) day to 7:00 a.m. on the following day) other than a Saturday, a Sunday or a holiday evening.
- 10.13 Employees who are required by Management to attend a training course, conference or seminar:
 - (a) Shall be paid their normal rate for a normal day only. Travel time or course time that is in addition to regular working hours will be accommodated within the Letter of Understanding re: Flexible Work Schedule Program Guidelines.
 - (b) Will be reimbursed for related expenses which are pre-approved in accordance with the applicable Corporate policies and procedures.

ARTICLE 11 – HOLIDAYS

11 .1 All employees within the scope of this Agreement who are not required to work on the following holidays shall be paid at the regular rate of pay for each of the following holidays:

| New Year's Day | Canada Day | Christmas Day |
|----------------|------------------|---------------|
| Good Friday | Civic Holiday | Boxing Day |
| Easter Monday | Labour Day | Lieu Day |
| Victoria Day | Thanksgiving Day | Family Day |

and any other day declared by a competent authority to be a holiday within the meaning of the <u>Bills of Exchange Act</u>. An employee in receipt of wage replacement benefits, not including Workplace Safety and Insurance Board Benefits, will receive the difference between the wage replacement benefit received and 100% of regular pay for paid holidays falling within the disability period. Such amounts to be payable upon return to work

11.2 In addition to the foregoing, the one-half (½) working day preceding Christmas Day and the one-half (½) working day preceding New Year's Day shall constitute an additional paid half holiday, provided however, the Christmas half (½) holiday and the New Year's half (½) holiday will be observed on the working day preceding the respective holiday except when Christmas Day and New Year's Day fall on a Saturday, Sunday or Monday, in which case the half (½) holiday will be observed on the preceding Friday afternoon.

- 11.3 Whenever any of the above holidays falls on a Saturday or Sunday, and are not proclaimed as being observed on some other day, the following Monday and/or Tuesday shall be deemed to be holidays for the purpose of this Agreement.
- 11.4 Subject to Article 11.3, employees who are required to work on any of the said holidays shall be paid in accordance with Article 10.6.

Where holidays are observed on days other than the calendar day on which they occur, employees who are required to work on the calendar day on which the holiday occurs shall receive any entitlement to holiday pay and the premium provided for in Article 10.6 for work performed on the calendar day, but not for any work performed on the date of observation.

- 11.5 When a paid holiday falls on one of an employee's regularly scheduled days off, he/she shall receive an alternate day off, with pay, on a day to be mutually agreed upon between the employee and the Managing Director or the appropriate Management supervisor, but in any event within no more than three (3) months after the holiday.
- 11.6 The one (1) lieu day as mentioned in Article 11.1 shall be afforded to the employee with seniority by the employee's Managing Director at a mutually agreeable time within the current calendar year.

ARTICLE 12 – VACATIONS

- 12.1 For the purpose of determining vacation entitlement the "vacation year" is defined as the twelve (12) month period starting July 1 and ending June 30 of the following year.
- 12.2 Employees who have less than one (1) year's seniority (as defined in Article 6.1) prior to the 1st day of July in a year shall be entitled to one (1) day's vacation with pay for each completed month of service up to a maximum of ten (10) working days vacation with pay. Vacation entitlement each year, per the attached table, is dependent on the years of service attained by June 30th in a year.

| 2 Weeks vacation | After 1 year |
|------------------|----------------|
| 3 weeks vacation | After 2 years |
| 4 weeks vacation | After 8 years |
| 5 weeks vacation | After 15 years |
| 6 weeks vacation | After 24 years |

- 12.3 An employee's vacation shall be taken in the twelve (12) months following the first of July in each year in which it is earned. A maximum of five (5) days may be carried over beyond the twelve (12) month period into the next year. The five (5) day maximum carryover may be exceeded with the approval of the applicable Managing Director. Such permitted postponed vacation time may be accumulated to a total of thirty (30) working days over a period of years. Notwithstanding this provision, no earned vacation shall be lost by an employee as a result of being unable to take same because of illness for which full salary is continuing from accumulated Sick Leave credits, or, as a result of a compensable accident, which illness or compensable accident does not permit the employee to complete their vacation allowance prior to the expiry of the twelve (12) months following which it is earned.
- 12.4 Employees who become hospitalized during an approved vacation, may substitute such sick leave as they have owing to them and will be granted alternative vacation days equivalent to the number of vacation days hospitalized (excluding non-scheduled work days) providing that:
 - (a) They were hospitalized in a recognized institution and verification of this is received by Management.
 - (b) The alternative days are taken at a time mutually convenient to the employee and Management.
- 12.5 No employee shall be permitted to forego their vacation period so that they may be paid in lieu of time off for vacation.
- 12.6 (a) Two (2) vacation planners per year as per chart outlined below will be introduced with Planner #1 covering the period June 1 to November 30 and Planner #2 covering the period December 1 to May 31. On or before the 5th day of March and September in each year, the Corporation will circulate a vacation planner so that employees may write in their choice of vacation dates. When preparing the semi annual vacation schedule, the Corporation shall, subject to its right to maintain a qualified working force, give the choice of vacation dates to employees with the greatest seniority within the work area or division as appropriate. The vacation schedule shall be completed on or before the 15th day of April and October as the case may be and when completed, copies shall be posted on the bulletin boards in the Departments or Divisions concerned for the information of the employees.

Circulate Planner

Planner #1 March 5 Planner #2 September 5

Deadline for Selection

Planner #1 April 1
Planner #2 October 1

Posting of Schedule

Planner #1 April 15 Planner #2 October 15

Periods Covered

Planner #1 June 1 to November 30 Planner #2 December 1 to May 31

- (b) If any employee fails to list their choice of vacation dates on or before April 1st, and October 1st as the case may be the Managing Director or the appropriate Management Supervisor shall require them to make a decision. Failing to promptly reply, the employee shall lose seniority preference for vacation purposes for that year.
- 12.7 In order to receive a pay advance for vacation, an employee shall make a request in writing to the Director of People Services not less than three (3) weeks in advance of the date the vacation is to commence. Such an advance shall cover only the pay period or pay periods falling within the employee's vacation.
- 12.8 In the event of the death of employees, their heirs or their estate shall be entitled to receive such vacation pay as may stand to the employee's credit.
- 12.9 Each person who has been employed by the Corporation on a temporary basis shall be paid vacation pay in accordance with the <u>Employment Standards Act, 2000</u> unless the Collective Agreement provides a greater benefit for temporary employees. In accordance with subsection 36(3) of the <u>Employment Standards Act, 2000</u> (as may be amended from time to time), the parties agree that the Corporation shall pay vacation pay for temporary employees that accrues during each pay period on the pay day for that period.
- 12.10 The vacation pay which is due to an employee under this Agreement shall not include overtime but shall include shift premium. Vacation pay shall be calculated upon the basis of annual salary on the date vacation commences.
- 12.11 Employees absent from the workplace, in receipt of Short Term Disability, Long Term Disability or Workplace Safety & Insurance Board benefits, for greater than twelve (12) consecutive months will not continue to earn vacation credits until such time as they return to work for the Corporation.

ARTICLE 13 - HOSPITAL, MEDICAL, SICK LEAVE, PENSIONS, GROUP INSURANCE

13.1 HEALTH AND WAGE LOSS INSURANCE PLANS

- (a) The Corporation will pay 100% of the premiums for the said health plans, as set out below:
 - The Ontario Health Insurance Plan.
 - The Manulife Supplementary (or equivalent coverage) to the Ontario Health Insurance Plan maximum prescription dispensing fee ten dollars (\$10.00) (unless maximum ODB dispensing fee is greater, and provide for voluntary generic substitution except on express instruction of physician).
 - Manulife Extended Health Care Benefits Plan, or equivalent coverage, which will include coverage for prescribed CPAP machines, no deductible will apply for single or family coverage. The plan will include the Deluxe Travel Plan.
 - Chiropractic, Osteopath, Naturopath, Podiatrist, Physiotherapist, Speech Pathologist, Masseur, Chiropodist, Social Worker and Psychologist coverage for a total of one thousand, five hundred dollars (\$1,500.00) per year.
 - Hearing Aid Coverage, no lifetime maximum.

Employees and dependents are encouraged to participate in the Health Care Partnership (HCP) to access reduced prescription fees as charged by the network pharmacies.

- (b) The Corporation will pay one hundred percent (100%) of the premiums for the dental plans, or equivalent coverage, as set out below. It shall be required of all employees electing such coverage to pay the balance of said premiums.
 - Manulife Dental Plan #9 with current Ontario Dental Association Fee Schedule Benefits.
 - Dental Rider #3 (orthodontics), on a 50/50 co-insurance basis with no deductible, and with a maximum lifetime benefit level of two thousand, five hundred dollars (\$2,500.00).
 - Pit and Fissure Rider and, Space Maintainers, no age limit.
 - Manulife Dental Plan Rider #4, (Major Restorative) on a 50/50 coinsurance basis with no deductible.

There is a nine (9) month recall provision for dental checkups for adults only (dependents excluded).

- (c) (i) The Corporation will pay one hundred percent (100%) of the premiums for a Vision Care Plan in conjunction with the Extended Health Care Plan providing for a benefit of three hundred and fifty dollars (\$350.00) in twenty-four (24) months with a twelve (12) month prescription rider plus loss or breakage replacement up to the maximum dollar amount. The Vision Care Plan benefit of three hundred and fifty dollars (\$350.00) in 24 months inclusive of laser eye surgery. Employees shall pay fifty percent (50%) of the premium of such plan; the five twelfths (5/12ths) portion of the E.I. premium reduction rebate for employees (respecting wage loss programs) shall be paid to the employer and shall be deemed to cover the employees' cost of this benefit, whether same is actually more or less than the rebate.
 - (ii) An eye exam once every twenty-four (24) consecutive months up to a maximum amount of ninety dollars (\$90.00) per exam for employees and eligible dependants.
- (d) (i) Enrolment in the benefit plans outlined in (a), (b), and (c) will be the first of the month following month of hire unless date of hire is prior to the 15th of the month in which case coverage will include the month of hire.
 - (ii) Existing dependent coverage shall be extended to include unmarried, unemployed dependant children over twenty-one (21) years of age, but under twenty-five (25) years of age in fulltime attendance at a school, college or university.
- (e) For employees retiring with twenty (20) (fifteen (15) for the term of the Collective Agreement) or more years of service and enrolled in the above insurance plans prior to retirement the Corporation will continue to provide coverage in the above insurance plans from date of retirement to age sixty-five (65).

For employees with a hire date after July 22, 2015 retiring with thirty (30) or more years of service and enrolled in the above insurance plans prior to retirement, the Corporation will continue to provide coverage in the above insurance plans from date of retirement to age sixty-five (65).

Appendix "A" attached to this Agreement outlines additional provisions for these retirement group insurances.

(f) The Ontario Drug Plan is an offset to drug coverage under the plan(s) of insurance and the annual deductible shall be covered by the plan(s) of insurance. Covered drugs are identified in the insurer's drug formulary (which may change from time to time).

13.2 SICK LEAVE

Employees in the bargaining unit shall be entitled to the sick leave and retirement gratuity benefits as per the following:

- (a) (i) Effective October 1, 1985, permanent employees with a seniority date prior to February 1, 1985, shall be eligible to a credit of one (1) day sick leave credit for each month of service with the Corporation. Such credits shall be cumulative.
 - (ii) Permanent employees with a seniority date on or after February 1, 1985, shall earn one (1) day (seven (7) hours) of sick leave credit for each complete month during which they worked all scheduled hours. (Provincial Offences Officers will earn one (1) day (eight point seven-five (8.75) hours). Vacation, Statutory Holidays, Bereavement leave, casual time off, time off in lieu of overtime, authorized Union Business or leave prescribed by the <u>Employment Standards Act, 2000</u> are considered hours worked. Such earned credits shall be cumulative.

Sick leave credits will accumulate to a maximum of eighty-five (85) days. [For employees hired after February 1, 1985, sick leave credits in excess of eighty-five (85) days may be used until the balance is reduced to the cap of eighty-five (85) days].

An absence (or absences) for any other reason shall mean the employee does not earn sick leave credits for that month, provided the cumulative total of such absences was greater than two (2) days (fourteen (14) hours or for Provincial Offences Officers seventeen point five (17.5) hours) or more.

- (b) Employees, after acquiring seniority with the Corporation shall be eligible to receive sick leave, at full salary or wage rate, for any time lost by illness to the full extent of Sick Leave Credits available to them at the time of such absence. Employees who will be absent from work for any reason, including reasons of illness for which they will be claiming the use of sick leave credits, are required to so advise the appropriate Management supervisor or designate at the beginning of each absence.
- (c) Except as otherwise herein provided, the number of days employees are absent on account of illness shall be deducted from their cumulative Sick Leave Credits

- (d) If their absence due to such sickness extends beyond a pay period, their pay shall, to the extent of their accumulated sick leave credits, be continued during such absence provided they give or cause the giving of prompt notice of their sickness.
- (e) (i) Employees who have been off work for illness or accident for five (5) days or more may be required to contact the Occupational Health Nurse or Physician during the first day upon return to work.
 - (ii) Employees who are absent because of sickness for three (3) days or more shall, on request, provide the Managing Director with a certificate from a qualified physician certifying as to their inability to return to work and on similar request, shall do so as the Managing Director may require. The employer shall pay 50% of the normal and customary fee, if applicable, for the certificate mentioned, or obtain a certificate from the Corporation physician.
- (f) Re-employed personnel of the Armed Forces shall receive the same Sick Leave Credit for the time spent in the Forces as they would have received had they remained with the Corporation.
- (g) Sick Leave Credit earned by service in any-Department shall be credited to the employee concerned, and sick leave pay to which the employee is entitled shall be authorized by the Department in which the employee is employed at the time of illness.
- (h) Employees with a seniority date prior to February 1, 1985, and who are, at the time of their retirement, actively engaged at their duties or absent on duly authorized leave, shall be entitled to receive a sick leave gratuity on one (1), but not both, of the following basis:
 - (i) On the date of retirement, such employees may be granted a sick leave gratuity in cash equal to their salary, wages or other remuneration for one-half (½) the number of days standing to their credit and in any event not in excess of the amount of one-half (½) year's earnings at the rate received by the employees immediately prior to termination of employment; or
 - (ii) With the consent of their Managing Director, in lieu of the sick leave gratuity which would otherwise be paid in cash in accordance with the foregoing, such employees may be granted retirement leave with full pay for a period equal to one-half (½) the number of days standing to their credit and in any event, not in excess of a period of six (6) months. Such leave shall be completed as of the date of normal retirement.

Employees with a seniority date on or after February 1, 1985, are not entitled to such benefit.

(i) Employees with a seniority date prior to February 1, 1985, and who on termination of their employment with the Corporation have at least seven (7) years of service; or the Estate of such employees who die while in the employ of the City, having at least seven (7) years service, shall be entitled to receive pay for the period equal to one-half (½) the number of days standing to their credit and, in any event not in excess of the amount of onehalf (½) year's earnings at the rate received by the employee immediately prior to termination of their employment.

Employees with a seniority date on or after February 1, 1985, are not entitled to such benefit.

- (j) (i) Whenever employees, formerly employed by another Ontario municipality or local board which had established a sick leave credit plan under the provisions of the *Municipal Act*, or any private Act of the Corporation of the City of London, become employees of the Council of the Corporation of the City of London, without interruption of employment by another employer, the Director of People Services shall ascertain, in writing, the extent of the sick leave credits, if any, standing to their credit in the plan of the Ontario municipality or local board formerly employing the employee, and the Director of People Services shall take such action as may be necessary to place such sick leave credits to the new employees credit in the records of the City of London.
 - (ii) Employees who give notice of absence due to sickness may be required to produce evidence of sickness reasonably satisfactory to the Director of People Services. In the event the Corporation requests an employee who is absent on sick leave to submit to a medical examination by a physician appointed by the Corporation, the Corporation shall be entitled to a copy of the physician's report.
 - (iii) Employees eligible for income indemnity or long term disability insurance do not earn sick leave credits for any time they were so eligible, unless the employee returns to active employment, at which time the employee would receive those credits which they would have otherwise earned.

13.3 PREVENTATIVE MEDICINE

The Corporation and the Union will co-operate in promoting a program of preventative medicine in order to reduce incidence of illness among employees.

13.4 PENSIONS

- (a) The Corporation and the Union agree to the Ontario Municipal Employee's Retirement System and the Canada Pension Plan as established.
- (b) Effective January 1, 1976, the Corporation agrees to provide an OMERS Supplementary Type II Pension Plan – two percent (2%) formula all service (past and future) updated to January 1, 1976 - normal retirement age of sixty-five (65) years with total cost paid by the Corporation. For purposes of clarity, this Plan will be provided for and will cover all employees who retire, or who have retired, subsequent to January 1, 1976 and prior to April 30, 1981.
- (c) Effective May 1, 1981, the Corporation agrees to provide an OMERS Type I Past Service Supplementary Pension Plan two percent (2%) formula normal retirement age of sixty-five (65) years, with the total cost paid by the Corporation. For purposes of clarity this Plan will be provided for, and will cover, all employees who retire subsequent to May 1, 1981 and will replace the OMERS Type II Supplementary Pension Plan referred to in Article 13.4(b) of this Agreement.

13.5 LIFE AND WAGE LOSS INSURANCE PLANS

(a) The Corporation will pay 100% of the premiums [except as otherwise noted below] for:

A Group Life Insurance Plan under which the life of each employee who is covered by this Agreement will be insured to the extent of two and one-half (2½) times an amount equal to the employee's annual salary calculated to the next one thousand dollars (\$1,000.00) up to a maximum of one hundred thousand dollars (\$100,000.00). The cost of the coverage will be one hundred percent (100%) paid by the Corporation.

An Accidental Death & Dismemberment (AD&D) Insurance Plan under which the life of each employee who is covered by this Agreement will be covered to the extent of two and one-half (2½) times an amount equal to the employee's annual salary calculated to the next one thousand dollars (\$1,000.00) up to a maximum of one hundred and twenty five thousand dollars (\$125,000.00). The cost of the coverage is one hundred percent (100%) paid by the Corporation.

Dependent life insurance in the amount of fifty thousand dollars (\$50,000.00), on the life of the spouse of the employee, and in the amount of twenty-five thousand dollars (\$25,000.00), on the life of each unmarried dependent child of the employee is optionally available to each participant at the entire premium cost of the employee concerned. The definition of dependent includes children to age twenty-one (21) or to age twenty-five

(25) if the child is in full time attendance at an educational institute. Election of coverage must occur within thirty-one (31) days of becoming eligible.

The disability waiver of premium benefit will be provided to age sixty-five (65), and the Corporation will pay one hundred percent (100%) of costs.

Employees will be insured under a Short Term Disability Plan which will provide a maximum benefit equal to seventy percent (70%) of an employee's regular weekly gross pay (amount equal to an employee's annual salary divided by fifty-two (52)) for new claims. Eligible employees who are medically disabled and unable to work will be entitled to weekly benefits commencing five (5) working days following the onset of the disability or upon the exhaustion of their accumulated sick leave credits pursuant to Article 13.2, whichever is later. The Short Term Disability benefit will be paid to a maximum of twenty-six (26) consecutive weeks for any individual medical disability which renders the employee unable to perform work.

The maximum benefit is eight hundred dollars (\$800.00) per week.

- (b) The Corporation shall pay an amount equal to one hundred percent (100%) of the premiums for the insurance mentioned in Clause (a) of this Article, except as otherwise noted. Participation in such group insurance plans shall be mandatory and shall be a condition of employment.
- (c) All permanent active employees, and new permanent employees upon completion of the probationary period will receive a Long Term Disability Insurance Plan under which insurance an employee who is sick and thereby unable to work will be paid seventy percent (70%) of gross regular earnings at the time the employee became sick; benefits under this said Long Term Disability Insurance shall commence one hundred ninety-six (196) days after the employee became sick, or upon the exhaustion of their accumulated sick leave credits pursuant to Article 13.2, together with any indemnity insurance benefits, whichever is later. The benefit shall continue until the earlier of recovery, death of the employee, or until the employee reaches age sixty-five (65).

The maximum monthly benefit is three thousand dollars (\$3,000.00). Further information in respect of this plan is set out in Appendix "B".

The Corporation shall pay one hundred percent (100%) of the premium cost of Long Term Disability Insurance.

(d) An employee who receives Short Term or Long Term Disability Plan Benefits will be paid the difference between those benefits and their regular daily pay for any of the paid holidays listed in the Collective Agreement which occur during a period of absence for which those benefits were paid.

- The aforementioned difference will be paid to the employee following their return to work
- (e) The Corporation shall continue the payment of the Employer's share of medical premiums in Article 13.1 for all employees in receipt of Short Term or Long Term Disability Benefits.
- 13.6 All of the insurance mentioned in this Article shall be as particularly described and set forth in the respective policy or policies of insurance. Any dispute over payment of benefits under any such policy or policies shall be adjusted between the insured or the beneficiary under such policy and the insurer concerned, but the Corporation will use its best efforts to adjust and settle any such dispute.
- 13.7 Every employee shall be fully responsible for keeping the City informed of changes in marital status or number of dependents. An employee who is entitled to reduced hospitalization or medical benefit premium, because of a change in their dependency status, and who fails to notify the City of such change, shall be responsible for the extra premium expense paid by the City on their behalf because of such failure to notify, and this extra cost shall be deducted from the employee's wages.
- 13.8 The Corporation will secure the policies outlined in this Agreement from the insurer(s), and the Corporation has the exclusive right to select such insurer(s), provided the insurer(s) maintain equal or better coverage.
- 13.9 For the purpose of Medical, Extended Health, Dental, and Life Insurance Coverage, spouse shall include same sex and common law spouse as defined by the Insurance Carrier.

ARTICLE 14 - REHABILITATIVE ASSIGNMENTS AND WORKPLACE SAFETY AND INSURANCE

- 14.1 The Corporation and the Union agree jointly to a workplace rehabilitation program aimed at returning employees with occupational or non-occupational disabilities or diminished capacity to gainful employment with the main objective to return the employee to the employee's regular pre-disability work.
- 14.2 The Corporation will provide, wherever practical, temporary rehabilitative work assignments to employees who may become temporarily disabled, and the Corporation and the Union agree that employees requiring rehabilitative work assignments will be given preference to such suitable work as is available and which they may be qualified to perform.
- 14.3 An employee to whom this Article applies shall be subject to an examination by the Corporation's Occupational Health Physician or another physician selected by the Corporation. If it is deemed appropriate for the employee to be assessed through an Independent Medical Examination, representatives of the Corporation and the Union will consult and agree in advance on expenses for expense sharing. The medical information

obtained through such an Independent Medical Examination will be provided only to the employee and to the Corporation's Occupational Health physician.

- 14.4 The employer shall make available to the Union through the Health and Safety Committee or upon request, copies of any Supervisor's Report of Injury relating to its members
- 14.5 Where employees are absent as a result of an accident while at work, or illness inherent to occupation, and, as a result, are receiving Workplace Safety and Insurance Board (WSIB) benefits, as awarded by the WSIB, they shall be paid by the Corporation the difference between their regular pay and the Board's award and the employees shall be paid such amount for the period of the award or until such time as the employees cease to draw salary from the Corporation. If such employees are not eligible for WSIB benefits, they shall receive sick leave pay according to this Agreement and the time off shall be deducted from their Sick Leave Credits.

Effective June 1, 1987, for new workers' compensation claims, where employees are absent as a result of an accident while at work, or illness inherent to occupation, and, as a result, is receiving workers' compensation, as awarded by the workers' compensation board, they shall be paid by the Corporation the difference between their regular net pay as calculated by the workers' compensation board and the board's award and the employees shall be paid such amount for the period of the award or until such time as the employee ceases to draw salary from the Corporation. Those payroll deductions not included in the calculation of regular net pay as calculated by the board will be deducted after the net is determined. Paid holidays specified in Article 11.1 occurring during the workers' compensation absence will accumulate for observance only upon return to work.

- 14.6 Employees applying for WSIB benefits will continue to be paid by the Corporation to the extent that sick leave credits are available to that employee, pending adjudication of their claim by the WSIB. Sick leave credits will be reinstated to the extent a claim is approved.
- 14.7 When an employee, due to disability, is placed in a position carrying a lower wage rate on a permanent basis, his or her existing wage rate shall be red-circled for a period of one (1) year from date of transfer. At the completion of this first year in the lower paid classification, his or her hourly rate shall be reduced by 3% and by a further 3% at subsequent six (6) month intervals. This retrogression shall proceed until the reduced wage rate and the wage rate of his or her new classification are the same. Upon reaching the rate of the lower paid classification, the employee concerned shall be granted negotiated increases for that classification.

If a lower rated position is not available and an employee, due to disability, is placed in a position carrying a higher wage rate of pay, the terms of Articles 9.5(a) and 9.4(a) shall apply.

ARTICLE 15 - GRIEVANCE PROCEDURE

15.1 (a) It is the mutual desire of the Corporation and the Union that all complaints and grievances shall be resolved as quickly as possible.

Notwithstanding Article 5.3, Stewards may have such time as is reasonably required during normal working hours to investigate grievances or employee complaints which may otherwise lead to grievances, and shall receive their regular pay for such absence, provided that permission is first obtained from their appropriate Management supervisor. Such permission will not be unreasonably withheld.

- (b) All meetings at which grievances are processed shall be held in camera.
- (c) Employees who are covered by this Agreement shall be required to follow the procedures laid down in this Article. Any employee who appeals directly to any elected official shall thereby forfeit all rights under this Article and under Articles 16 and 17.
- (d) A grievance under this Agreement shall be defined as any difference or dispute between the Corporation and any employee or the Union relating to the interpretation, application or administration of this Agreement, including any question as to whether a matter is arbitrable and an allegation that this Agreement has been violated.
- (e) The time limits fixed in this Article and in Article 16 may be extended by consent of both parties to this Agreement.
- 15.2 In the event of a grievance of an employee, the employee shall take the matter up within and not after five (5) working days after the employee became aware or ought reasonably to have become aware of the incident or circumstances giving rise to the grievance. A policy grievance or group grievance shall be taken up within and not after ten (10) working days after the date of the incident giving rise to the grievance or the aggrieved party ought reasonably to have become aware of such incident or circumstances.
- 15.3 The following is the procedure which shall be adhered to in processing grievances, save as otherwise provided in this Article and in Article 17:

Step No.1

The employee shall discuss the matter with their appropriate Management Supervisor and an attempt shall be made to adjust the employee's complaint. At the request of either the appropriate Management supervisor or the employee, one member of the Union Grievance Committee shall be present during such discussion.

Step No.2

If the complaint is not settled within and not after five (5) working days of the date of discussing the complaint with the employee's appropriate Management Supervisor (or if no decision is received from the employee's appropriate Management Supervisor within five (5) working days after such decision ought to have been given), the complaint shall be reduced to a written grievance and the aggrieved employee, and one member of the Union Grievance Committee, may formally submit the grievance to the Director of People Services or designate. The Director of People Services or designate, who may have the assistance of the employee's Managing Director, or their nominee as required, shall take the matter up with the Grievor and the Union Grievance Committee member within ten (10) working days after the date the grievance is formally submitted to the Director of People Services or designate. The Director of People Services or designate, within ten (10) working days of meeting with the Grievor and the Union Grievance Committee member, shall advise the Union in writing of the Employer's response. If the grievance is not settled within ten (10) working days from the date of the meeting, the Union may, within and not after twenty (20) working days after the date it receives the Employer's written response (or if no written response from the Director of People Services or designate is received within ten (10) working days, within and not after twenty (20) working days after such response ought to have been given), refer the grievance to Arbitration under Article 16.

- 15.4 (a) A grievance filed by a group of employees and a policy grievance of the Union shall be taken up at Step 2 of the grievance procedure.
 - (b) A policy grievance of the Corporation shall be in writing and may be initiated by the Director of People Services or designate delivering the grievance to the President of the Union, or, in his/her absence to another officer of the Union. If any such grievance is not settled within fifteen (15) working days of the date of such delivery the Corporation may refer the grievance to Arbitration under Article 16.
 - (c) An interdepartmental grievance filed by an employee shall be taken up at Step 2 of the grievance procedure.

ARTICLE 16 - ARBITRATION

16.1 Where a difference arises between the parties relating to the interpretation, application or administration of this Agreement, including any question as to whether a matter is arbitrable, or where an allegation is made that this Agreement has been violated, either of the parties may, after duly exhausting the grievance procedure established by this Agreement, notify the other party in writing of its desire to submit the difference or allegation to Arbitration and the notice shall contain the name of the first party's appointee to an Arbitration Board. The recipient of the notice shall, within five (5) days inform the other party of the name of its appointee to the Arbitration Board. The two appointees so selected shall proceed to appoint a third person who shall be the Chairperson. If the recipient of the notice fails to appoint an arbitrator, or if the two

appointees fail to agree upon a Chairperson within fifteen (15) days, the appointment shall be made by the Ontario Ministry of Labour Office of Arbitration upon the request of either party. The Arbitration Board shall hear and determine the difference or allegation and shall issue a decision and the decision is final and binding upon the parties and upon any employee affected by it. The decision of a majority is the decision of the Arbitration Board, but if there is no majority the decision of the Chairperson governs. The Arbitration Board shall not have any authority to alter or change any of the provisions of this Agreement or to substitute any new provision in lieu thereof or to give any decision contrary to the terms and conditions of this Agreement, or in any way modify, add to or detract from any provision of this Agreement. Each of the parties to this Agreement will pay the fees and disbursements of its appointee to the Arbitration Board and will share equally the fees and disbursements of the Chairperson.

16.2 Where both parties agree, a single arbitrator with the same limitation and powers as a Board of Arbitration may be substituted for a Board of Arbitration, except it is understood that either party may apply for a sole Arbitrator under the "expedited arbitration" provisions of the <u>Ontario Labour Relations Act</u>.

ARTICLE 17 - DISCHARGE AND DISCIPLINE CASES

- 17.1 In the event an employee who has attained seniority is discharged or disciplined and the employee considers that an injustice has been done, the matter may be taken up at Step 2 of the grievance procedure.
- 17.2 (a) If an employee is suspended without pay pending investigation, a decision will be made by Management within three (3) days to confirm or cancel the suspension or to adjust the length of suspension, except that in extenuating circumstances when a decision cannot be reached within three (3) days, Management and the Union will meet to determine whether or not the employee shall be returned to work during a period of further investigation.
 - (b) If the suspension is cancelled, the employee shall receive pay for all normal time lost. If the suspension is found to be valid, the date of suspension shall be retroactive to the first day of suspension.
 - (c) Management shall not suspend or discipline an employee without a steward being present, except in circumstances where an employee is suspended pending investigation, in accordance with 17.2 a) and b). In the latter case, the Union shall be informed as soon as practicable and in no event later than one working day after the commencement of the suspension.
- 17.3 Where an employee's grievance against his/her discharge or discipline duly comes before an Arbitration Board, the Board may make a ruling:
 - (a) Confirming the Corporation's action, or

- (b) Reinstating the employee with or without compensation for wages lost (except for the amount of any remuneration the employee has received elsewhere pending the disposition of his/her case), or
- (c) Disposing of the grievance in any other manner which may be just and equitable.
- 17.4 Management may discipline or dismiss a probationary or other non-seniority employee for any bona fide reason provided it does not act in bad faith and this shall constitute a lesser standard for the purposes of section 43.1 of the <u>Ontario Labour</u> Relations Act.

ARTICLE 18 - PROTECTION OF EMPLOYEE'S POSITION

18.1

- (a) The Corporation agrees that it will not put out for tender or contract or employ any person or persons or group of persons for any job now filled by any employee falling within the scope of this Agreement, so as to have the effect of depriving employees covered by this Agreement of their employment.
- (b) Persons who are not in the bargaining unit shall not work in any jobs which are included in the bargaining unit, so as to have the effect of depriving employees covered by this Agreement of employment, except as mutually agreed upon by the parties.

18.2 Technological Changes and Reorganization

- (a) When the Corporation is considering introducing significant technological changes or reorganizations such that either the employment status or pay classification of employees covered by this Agreement will be affected, the Corporation agrees to notify the union as far in advance as possible providing known details of the change and the foreseeable effects and repercussions on employees. Thereafter the employer agrees to notify affected employees. Such information will be updated as new developments arise or modifications to plans are made.
- (b) Training Benefits where, as the result of such change, new or greater skills are required than previously required, the Parties recognize the mutual responsibility both of affected employees and the employer to ensure that skills upgrading is undertaken. To the extent practical in consideration of operating efficiencies, and giving special consideration to the senior affected employees, the Corporation will provide employees with opportunities during working hours to develop such skills within a reasonable time frame. The Corporation will make reasonable efforts to develop a skills upgrading plan which may include external courses of instruction.

ARTICLE 19 - CAR AND CLOTHING ALLOWANCE

19.1 (a) Employees required to use their own cars on City business shall receive a mileage allowance in accordance with the Corporate policy in place from time to time or forty cents (\$0.40) per km, whichever is greater

The Corporation shall designate the employees who are required to use their cars as aforesaid provided that upon one month's written notice to that effect, an employee will no longer be required to use their car on the Corporation's business and at the end of that month, car allowance will cease.

Employees called into work, outside of business hours, will be paid mileage calculated from their place of work to the work assignment destination.

- (b) Employees required to use their own vehicle for Corporation business will be reimbursed for parking expenses incurred upon submission of proper receipts.
- 19.2 The Corporation agrees to provide at its own expense to active employees an issue of clothing and safety equipment as set out in Schedule "B" of this Agreement.
- 19.3 The Joint Clothing Committee will determine the nature and specifications of such clothing and safety equipment.
- 19.4 Summer clothing will be issued on or before the first day of May and winter clothing on or before the first day of October.
- 19.5 The Corporation will pay one hundred dollars (\$100.00) per year, on a pro rata basis per month, to employees designated by the Corporation to carry heavy, dirty or bulky equipment in their personal automobiles while engaged in the Corporation's business
- 19.6 The Corporation will provide at its expense prescription glasses for employees required to operate video display terminals, provided such glasses are prescribed, monofocular, and ground for the specific focal length required for such use. The Corporation's cost is restricted to the cost of the lens, and basic frames.

ARTICLE 20 - UNION CONFERENCE, CONVENTION OR SEMINAR

- 20.1 Leave of absence with pay and without loss of seniority shall be granted to not more than three (3) employees who are elected or appointed to represent the Union at a conference, convention or seminar which may include other conference, convention or seminar related Union business.
- 20.2 An employee who is elected or appointed to a full-time elected position with the Union shall be granted leave of absence without pay and without the other benefits

provided by this Agreement, but without loss of seniority, for a maximum period of two (2) terms in office, which terms shall not exceed four (4) consecutive years.

An employee who is selected or appointed to a full-time position with the Union shall be granted up to twelve (12) months leave of absence without pay and without the other benefits provided by this Agreement, but without loss of seniority.

While on such leave of absence, the employee may make across-the-counter payments to continue their medical, hospital, pension and other benefits under this Agreement.

Upon the expiration of either of such types of leave of absence, employees shall resume duties with the Corporation, or shall be considered to have tendered resignation from the employ of the Corporation if they fail to return at that time.

ARTICLE 21 - LEAVES OF ABSENCE

- 21.1 Employees who are summoned to serve as a juror or are required by writ of subpoena to appear in court as a witness will be paid their regular pay for the time the employee is required to be in court, provided they present to the Corporation the process which required their presence in court and pays over to the Corporation the amount received by the employee as such juror or witness.
- 21.2 A Managing Director may grant leave of absence without pay in case of personal emergency.
- 21.3 The Corporation will provide pregnancy and parental leaves in accordance with the *Employment Standards Act*, Province of Ontario.
- 21.4 Leave of absence shall be granted with pay to an employee required to attend a formal Hearing to become a Canadian citizen.
- 21.5 At the discretion of the Managing Director, employees shall be allowed leave of absence without pay and without loss of seniority to upgrade their education.

Notwithstanding the provisions of the Collective Agreement, group insurance benefits, service accumulation, the payment of holiday pay, and the earnings of vacation credits shall cease thirty (30) days after the commencement of an Educational Leave and will not resume until the employee returns to active employment.

ARTICLE 22 - BEREAVEMENT LEAVE

- 22.1 Two (2) members of the Union shall be given time off, with pay, to attend funerals of City employees.
- 22.2 (a) In the case of the death in the immediate family, namely: spouse, child, step-child, mother, step-mother, father, step-father, sister, brother, step-father, sister, brother, step-father, sister, brother, step-father, step-fathe

- sister, step-brother an employee shall be permitted to be absent for not more than five (5) days with pay.
- (b) In the case of the death of a mother-in-law, father-in-law, grandmother, grandfather, or grandchild an employee shall be permitted to be absent for not more than three (3) days with pay.
- (c) For a relationship beyond that such as uncles, aunts, cousins, nephews nieces, sisters-in-law and brothers-in-law, spouse's grandparents, spouse's aunt, spouse's uncle, the employee shall be permitted to be absent for one (1) day with pay.
- (d) Where special circumstances arise, the time limits in this section (22.2) may be extended by up to two (2) days at the discretion of the Managing Director.
 - In the case of the death of a child or step-child, the time limits in section 22.2(a) may be extended by up to ten (10) days with pay at the discretion of the Managing Director.
- (e) If bereavement leave occurs during an employee's scheduled vacation or an employee's illness, time off will constitute bereavement leave pursuant to Article 22 and any vacation or sick leave credits used will be re-instated to the extent that Article 22 would allow.
- (f) Spouse will be deemed to include a common law spouse or same sex spouse.

ARTICLE 23 - TEMPORARY ASSIGNMENT

- 23.1 The provisions of this Article apply when the Corporation proposes an appointment to a temporary assignment, by filling:
 - (a) A permanent position on a temporary basis, or
 - (b) A temporary position.

It is agreed that the Corporation has the right to do either, in accordance with this Article.

- 23.2 A position is:
 "temporary" if it has an anticipated duration of twenty-four (24) months or less.
- 23.3 An employee is:
 "temporary" if not entitled to seniority pursuant to this Agreement.
- 23.4 Notice of a temporary assignment which has an anticipated duration of more than fifteen (15) weeks and not more than twenty-four (24) months shall be posted.

Notwithstanding the Corporation's right to determine a temporary assignment for a permanent position for greater than 24 months, notices of temporary assignments of more than twenty-four (24) months may be posted subject to mutual agreement of the Corporation and the Union. Such notice shall state the anticipated duration of the temporary assignment. The posting will indicate whether it is a permanent or temporary position being filled by temporary assignment.

- 23.5 Applications from permanent employees will be considered for posted temporary assignments in accordance with Article 8.2(a) and if successful the applicant shall be subject to the trial period outlined in Article 8.3(a). It is noted that a permanent employee, appointed to a temporary assignment, subsequently successful for a second temporary assignment will give up all rights to the first temporary assignment.
- 23.6 The Corporation may hire a person to fill a temporary assignment:
 - (a) If no permanent employee is appointed pursuant to the notice of temporary assignment.
 - (b) If no notice of temporary assignment is required.
- 23.7 A temporary assignment which has not been posted shall not be continued past fifteen (15) weeks, or revived within six (6) weeks of its expiry. The Corporation will notify the Union of all appointments of fifteen (15) weeks or less.
- 23.8 During the first thirty (30) weeks of employment in a temporary assignment, a temporary employee shall be paid in accordance with Schedule "A", and have the protection of Articles 2, 10, 11 and 17.4 of the Collective Agreement except as noted in sub-article 23.10. No other Article of the Collective Agreement applies. A temporary employee shall not be entitled to a lieu day holiday and shall, as a condition for other paid holidays, have worked the last working day before the holiday and the first regular working day after the holiday.

Temporary employees shall be paid vacation pay in accordance with the *Employment Standards Act*, 2000 and in accordance with subsection 36(3) of the *Employment Standards Act*, 2000, (as may be amended from time to time), the parties agree that the Corporation shall pay vacation pay for temporary employees that accrues during each pay period on the pay day for that period.

- 23.9 After the first thirty (30) weeks in a temporary assignment a temporary employee shall be covered by all terms and conditions of the Collective Agreement, except as noted in sub-article 23.10. Benefits once earned in accordance with this thirty (30) week provision will continue, during any probationary period, providing there is no break in service.
- 23.10 The following limitations apply to temporary employees:
 - (a) No seniority is earned.

- (b) The Corporation may, in its discretion, layoff or terminate the employment of a temporary employee:
 - Except that in the event of a disciplinary termination Article 17.4 shall govern.
 - (ii) Without notice in the case of a temporary assignment which has not posted.
 - (iii) Upon five (5) working days' notice in the case of a temporary assignment which has been posted and to which sub-article 23.8 applies.
 - (iv) Upon ten (10) working days' notice in the case of a temporary assignment which has been posted and to which sub-article 23.9 applies.
- (c) A temporary employee is not entitled to "bump" and may be bumped in the event of lay-off.
- (d) It is specifically noted that a temporary employee is not eligible to participate in the Ontario Municipal Employees Retirement System (OMERS), except as prescribed by OMERS Act and regulations.
- (e) A permanent employee who bumps into a temporary position becomes a temporary employee governed by sub-article 23.8 except that:
 - (i) Membership in OMERS must be continued.
 - (ii) Accumulated sick leave credits are frozen until seniority is restored or employment terminates, and;
 - (iii) Participation in the hospital and medical plans available to eligible employees may be continued at the employee's option and expense.
- 23.11 A temporary employee shall be deemed to be a permanent employee if the employee is the successful applicant for a permanent position and successfully completes the probationary period outlined in Article 6.2, the employee shall thereafter be covered by all terms and conditions of the Collective Agreement with a deemed seniority date corresponding to the date of last hiring (employment shall not be considered interrupted by a lay-off of less than eleven (11) working days) for the purpose of vacation credits, layoff, promotion, and sick leave credits only.
- 23.12 After twenty-four (24) months duration, a temporary position, excluding permanent positions filled on a temporary basis, shall be deemed permanent and will be posted in accordance with Article 8 of the Collective Agreement.

ARTICLE 24 - PART TIME EMPLOYMENT

- 24.1 The Corporation may hire no more than four (4) part-time employees per Department without the written mutual agreement of the Parties subject to the following provisions:
 - (a) All part-time vacancies shall be posted in accordance with the provisions of Article 8 of this Agreement.
 - (b) All terms of this Agreement shall apply to successful applicants for part-time vacancies except that benefits, seniority, vacation and progression on the salary grid shall be pro rata based on hours worked annually.
- 24.2 A part time employee is defined as an employee who is regularly scheduled to work up to twenty-one (21) hours per week and, may occasionally work more than twenty-one (21) hours per week to facilitate certain jobs. In no case will a part time employee work more than eighty-four (84) hours in two (2) consecutive pay periods.
- 24.3 (a) Notwithstanding Article 24.1(b) of the Collective Agreement, the parties agree that part time employees shall accumulate seniority based on: regular hours worked, paid vacation time, paid lieu day time, paid statutory holiday time, paid sick time, paid bereavement and paid flex time.
 - (b) In addition, part-time employees shall accumulate seniority while in receipt of Short Term Disability benefits, Long Term Disability benefits and Workplace Safety Insurance Act benefits, while on a personal leave of absence without pay for thirty (30) days or less, while on pregnancy and/or parental leave and any other leave provided for under the <u>Employment Standards Act 2000</u>. During these time periods, part-time employees shall accumulate seniority based on an average number of paid hours (as described in paragraph (a) above) per week calculated over the six (6) month period directly prior to the first day of absence.
 - (c) Notwithstanding Article 6, a part-time employee's seniority will be expressed in hours.

ARTICLE 25 - STUDENT EMPLOYMENT

25.1 The Union agrees that the Corporation has the right to employ Co-op Students under the Co-op Student Program as recognized in Ontario Universities and Colleges throughout the calendar year.

The Union also agrees that the Corporation has the right to employ other students during the period from May 1 through September 30, inclusive, of each calendar year.

A student hired during this period is defined as a student enrolled full time in a recognized college or university program and who intends to return to school following his/her term of

employment and/or the co-op placement is the final component prior to graduation.

- 25.2 The rate of pay for a student shall be according to Schedule "A"; students shall not be covered by any of the other terms and conditions of this Agreement, save for Article 2, 10, 11 (not including the lieu day, and provided the employee is at work on the last working day before and the first working day following, each of the other named paid holidays) and 12.
- 25.3 In the event the Corporation decides to reduce the numbers of the permanent staff complement, or, in the event the Corporation decides to not fill a vacancy or vacancies occurring in a permanent or temporary position, it is agreed that such vacancy shall not be filled by the hiring of a Co-op or other student.

ARTICLE 26 - NO STRIKES OR LOCK-OUTS

- 26.1 So long as this Agreement continues to operate, there will be no strikes or lockouts as those terms are defined in *The Labour Relations Act*.
- 26.2 In the event person(s) who are not covered by this Agreement engage in a lawful strike or are locked out at the Corporation of the City of London, members who are covered by this Agreement shall not be required to perform work normally done by the person(s) who are on strike or locked out.

ARTICLE 27 - EDUCATION ALLOWANCE

- 27.1 The Corporation shall reimburse employees one hundred percent (100%) of the tuition cost of a course of instruction taken by such employees to better qualify them to perform their jobs. Such courses must be approved in writing by the employee's Managing Director prior to commencement. Reimbursement will be upon receipt of evidence of successful completion of the course and tuition paid.
- 27.2 If employees wish to take time off from work for attendance at a University, College or Institute to take a course of instruction oriented towards their work, the Managing Director concerned may grant such time off providing the employees shall make the time up on an hour for hour basis, and further provided that such attendance will not interfere with the regular operation of the Department. Such time shall be made up on a day and time that is mutually agreed to by both the employee and the Manager. If it is not possible for the employees to make up the time, they shall be paid on a pro-rata basis for the actual hours worked. Payment for such courses shall be in accordance with Article 27.1.
- 27.3 The Corporation will pay the annual fees for all required licenses and professional memberships.

ARTICLE 28 - LAY-OFF AND RECALL

- 28.1 Whenever a position is declared redundant or the Corporation proposes to reduce the work force without declaring positions redundant, the Corporation shall immediately notify the Union of its intention at the same time it provides the incumbent(s) with the notice of lay-off. The Corporation shall notify employees subject to layoff in accordance with the *Employment Standards Act*.
- 28.2 If the Corporation proposes to reduce the work force without abolishing positions, the Corporation shall give the incumbents of the positions which it proposes to make vacant notice of lay-off.
- 28.3 (i) An employee who receives notice of lay-off may by notice within five (5) working days to the Director of People Services "bump" (displace) a less senior employee who occupies a position of equal or lower classification, provided that the employee has the necessary skill, ability and qualifications to perform the duties involved. No trial or training period shall be allowable on a bump but it is understood that any assessment shall be done in good faith and on a reasonable basis. A bumped (displaced) employee shall be given notice of lay-off and shall have the right to bump another employee on these same terms.
 - (ii) Notwithstanding the above, an employee who is filling a temporary position when that employee's permanent position is declared redundant, shall select the position they wish to "bump" into but shall remain in their current temporary position until its term is completed.
- 28.4 Recall shall be on the basis of seniority, qualifications, skill and ability. No new employee shall be hired while a qualified, skilled and able employee remains on lay-off.
- 28.5 Employees who are laid off will be retained on the seniority list for a period of twelve (12) months. If, during that period, they are recalled to work, they must signify their intention to do so within three (3) days (excluding Saturdays, Sundays and paid holidays) of the date of the notice of recall, and shall, in fact, return to work within a further five (5) days, or they shall forfeit their claim of employment and be deemed to have terminated their services. Employees have the right to decline a recall and shall remain on the seniority list with the same recall rights until the expiry of the twelve (12) months from date of initial lay-off. Notice of recall shall be given by the Corporation by registered mail to their last address on record with the Human Resources Division. The delivery date record of the post office will be the determining date with respect to giving notice of intention to comply with the recall request noting that no more than two (2) weeks are to be allowed for receiving registered mail calculated from the date on which such letter is sent.
- 28.6 Any employee who has completed their probationary period and is laid off, may continue to participate in the hospital and medical plans available to the employee for a period of up to twelve (12) months, provided that the employee pays to the City Treasurer, the full premium cost of such participation, not later than the twenty-fifth (25th)

day of the month prior to the month of which the payment becomes due. If such payment is not made as aforesaid, the employee's participation in these Plans shall be terminated forthwith. An employee who is recalled will be reinstated in the hospital insurance and medical plans immediately upon return to work. The premiums for all of the plans will be paid effective the first day back at work as if the employee had not been on lay-off. Premiums which were paid in advance by the employee will be recalculated so that the Corporation will pay on a pro rata basis for the period in the month during which the employee paid the full premium. The employee will be refunded the difference arising from the recalculation.

- 28.7 When temporary or probationary employees are returned to work in their former position after a layoff of not more than ninety (90) calendar days, they shall be paid at a rate not less than the rate they were receiving prior to layoff.
- 28.8 When employees are to be laid off, they shall be allowed reasonable time off during their last shift in order to attend to any personnel or pay related matters not yet settled.

ARTICLE 29 – GENERAL

- 29.1 Either Party to this Agreement may, within the period of ninety (90) days before the Agreement ceases to operate, give notice in writing to the other Party of its desire to bargain with a view to the renewal, with or without modification, of this Agreement. Within fifteen (15) working days of receipt of such notice, or within such further period as the Parties agree upon, they shall bargain in good faith and make every reasonable effort to arrive at a Collective Agreement.
- 29.2 The Union and the Corporation desire every employee to be familiar with the provisions of this Agreement and its rights and duties under it. Accordingly, true copies of this Agreement (together with all schedules and appendices) shall be printed in a Union shop and bear the appropriate Union labels. One-half ($\frac{1}{2}$) the cost of such printing shall be paid by the Corporation and one-half ($\frac{1}{2}$) shall be paid by the Union.

ARTICLE 30 - TERMS AND DEFINITIONS

POSITION - A collection of tasks, duties and responsibilities regularly assigned to and performed by a single individual.

JOB - A group or range of duties or tasks assigned to and performed by one or more individuals.

ARTICLE 31 - TERM OF AGREEMENT

This Agreement shall be for a term of forty eight (48) months, commencing on the 1st day of January, 2019, and ending the 31st day of December 2022, and thereafter in each succeeding year, subject to changes and amendments agreed to by both Parties in Writing.

IN WITNESS WHEREOF the Parties have executed this Agreement:

| 1 () | _ |
|----------------------|-----------------------|
| FOR THE UNION: | |
| President | NN.13 2019 Date |
| Secretary | NOV. 13, 2019 Date |
| FOR THE CORPORATION: | |
| Mayor | DEC 1 6 2019 |
| Duds | DEC 1 6 2019 |
| City Clerk | Date |

| HB |
|----------------|
| 2019 M ONLY |
| |

SCHEDULE A - POSITION CLASSIFICATION AND SALARY SCHEDULE Effective January 1, 2019

Department Descriptions:

City Manager's Office - CMO

Community Services – CS
Environmental and Engineering Services - EES

Financial and Corporate Services - FCS

Planning Tourism London

| Code Code | Position Title | Department | Year | Start | 6 Months | 1 Year | 18 Months | 2 Years | 30 Months |
|-----------|-------------------------------|------------|----------|----------|-------------|-----------|--------------|------------|--------------|
| C0100 | | | 1/1/2019 | \$25,343 | \$27,796 | \$30,422 | \$31,713 | \$32,997 | \$34,855 |
| | | | 1/1/2020 | \$25,787 | \$28,282 | \$30,954 | \$32,268 | \$33,574 | \$35,465 |
| | | | 1/1/2021 | \$26,264 | \$28,805 | \$31,527 | \$32,865 | \$34,195 | \$36,121 |
| | | | 1/1/2022 | \$26,789 | \$29,381 | \$32,158 | \$33,522 | \$34,879 | \$36,843 |
| Class | Position Title | Department | Year | Start | 6 Months | 1 Year | 18 Months | 2 Years | 30 Months |
| C0319 | Students (No evaluation) | n/a | 1/1/2019 | \$29,146 | \$32,067 | \$35,291 | \$36,810 | \$38,358 | \$41,159 |
| C0355 | Community Services Clerk | SO | 1/1/2020 | \$29,656 | \$32,628 | \$35,909 | \$37,454 | \$39,029 | \$41,879 |
| C0356 | Traffic Survey Clerk | EES | 1/1/2021 | \$30,205 | \$33,232 | \$36,573 | \$38,147 | \$39,751 | \$42,654 |
| C0361 | Co-op Student (No evaluation) | | 1/1/2022 | \$30,809 | \$33,897 | \$37,304 | \$38,910 | \$40,546 | \$43,507 |

| Class | | | | | c | - | 6 | œ | 4 |
|-------|--|-------------------|----------|----------|----------|----------|----------|----------|----------|
| Code | Position Title | Department | Year | Start | Months | Year | Years | Years | Years |
| C0425 | Administrative Assistant | EES | 1/1/2019 | \$31,053 | \$34,469 | \$38,293 | \$40,213 | \$42,159 | \$44,311 |
| C0453 | Divisional Administrative Assistant | EES | 1/1/2020 | \$31,596 | \$35,072 | \$38,963 | \$40,917 | \$42,897 | \$45,086 |
| C0462 | Custodian | EES | 1/1/2021 | \$32,181 | \$35,721 | \$39,684 | \$41,674 | \$43,691 | \$45,920 |
| C0463 | Administrative Assistant – Sports | SS | 1/1/2022 | \$32,825 | \$36,435 | \$40,478 | \$42,507 | \$44,565 | \$46,838 |
| C0464 | Divisional Administrative Assistant | EES | | | | | | | |
| C0479 | Administrative Support Clerk | FCS | | | | | | | |
| C0483 | Casework Unit Clerk | SS | | | | | | | |
| C0484 | Clerk Assessment | FCS | | | | | | | |
| C0487 | Seasonal Tourism Counsellor | Tourism London | | | | | | | |
| C0488 | Accounts Processing Clerk | EES | | | | | | | |
| C0491 | Records Clerk | FCS | | | | | | | |
| C0492 | Administrative Assistant | SS | | | | | | | |
| C0493 | Eligibility Review Administrative Assistant | cs | | | | | | | |
| C0494 | Municipal Election Clerk | FCS | | | | | | | |

| Code | Position Title | Department | Year | Start | 6 Months | 1 Year | 2 Years | 3 Years | 4 Years |
|-------|---|-------------------|----------|----------|-------------|-----------|------------|------------|------------|
| C0501 | Divisional Administrative Assistant – Water Engineering | EES | 1/1/2019 | \$32,926 | \$36,653 | \$40,761 | \$42,854 | \$44,951 | \$47,465 |
| C0503 | Tax Transfer Clerk | FCS | 1/1/2020 | \$33,502 | \$37,294 | \$41,474 | \$43,604 | \$45,738 | \$48,296 |
| C0527 | Divisional Administrative Assistant | EES | 1/1/2021 | \$34,122 | \$37,984 | \$42,241 | \$44,411 | \$46,584 | \$49,189 |
| C0542 | Forestry Clerk/Dispatcher | EES | 1/1/2022 | \$34,804 | \$38,744 | \$43,086 | \$45,299 | \$47,516 | \$50,173 |
| C0547 | Divisional Administrative Assistant | EES | | | | | | | |
| C0558 | Divisional Administrative Assistant | EES | | | | | | | |
| C0564 | Customer Service Representative - Planning | Planning | | | | | | | |
| C0566 | Divisional Administrative Assistant | EES | | | | | | | |
| C0574 | Support Clerk – Building | Planning | | | | | | | |
| C0576 | Administrative Assistant | cs | | | | | | | |
| C0577 | Communications, Research & Membership Clerk | Tourism London | | | | | | | |
| C0578 | Address Book Clerk | FCS | | | | | | | |
| C0580 | Public Service Clerk | EES | | | | | | | |
| C0581 | Administrative Assistant | cs | | | | | | | |

| Code | Position Title | Department | Year | Start | 6 Months | 1 Year | 2 Years | 3 Years | 4 Years |
|-------|--|-----------------------|----------|----------|-------------|-----------|------------|------------|------------|
| C0584 | Divisional Administrative Assistant | EES | 1/1/2019 | \$32,926 | \$36,653 | \$40,761 | \$42,854 | \$44,951 | \$47,465 |
| C0585 | Administrative Assistant – Payroll | FCS | 1/1/2020 | \$33,502 | \$37,294 | \$41,474 | \$43,604 | \$45,738 | \$48,296 |
| C0592 | Divisional Administrative Assistant | EES | 1/1/2021 | \$34,122 | \$37,984 | \$42,241 | \$44,411 | \$46,584 | \$49,189 |
| C0595 | Administrative Assistant – Finance | FCS | 1/1/2022 | \$34,804 | \$38,744 | \$43,086 | \$45,299 | \$47,516 | \$50,173 |
| C0596 | Administrative Assistant | CS | | | | | | | |
| C0597 | Research Assistant | CS | | | | | | | |
| Class | Position Title | Department | Year | Start | 6 Months | 1 Year | 2 Years | 3 Years | 4 Years |
| C0612 | Collections & Payable Clerk | SS | 1/1/2019 | \$34,846 | \$38,842 | \$43,296 | \$45,549 | \$47,804 | \$50,620 |
| C0639 | Corporate Services Assistant | Corporate Services | 1/1/2020 | \$35,456 | \$39,522 | \$44,054 | \$46,346 | \$48,641 | \$51,506 |
| C0652 | Administrative Assistant | EES | 1/1/2021 | \$36,112 | \$40,253 | \$44,869 | \$47,203 | \$49,541 | \$52,459 |
| C0664 | Divisional Administrative Assistant | EES | 1/1/2022 | \$36,834 | \$41,058 | \$45,766 | \$48,147 | \$50,532 | \$53,508 |
| C0665 | Sports Tourism Assistant | Tourism London | | | | | | | |
| C0666 | Senior Tourist Counsellor | Tourism London | | | | | | | |

| Code | Position Title | Department | Year | Start | 6 Months | 1 Year | 2 Years | 3 Years | 4 Years |
|-------|---|-------------------|----------|----------|-------------|-----------|------------|------------|------------|
| C0667 | Conventions Tourism Assistant | Tourism London | 1/1/2019 | \$34,846 | \$38,842 | \$43,296 | \$45,549 | \$47,804 | \$50,620 |
| C0668 | Leisure Tourism & Advertising Assistant | Tourism London | 1/1/2020 | \$35,456 | \$39,522 | \$44,054 | \$46,346 | \$48,641 | \$51,506 |
| 69900 | Lab Clerk | EES | 1/1/2021 | \$36,112 | \$40,253 | \$44,869 | \$47,203 | \$49,541 | \$52,459 |
| C0670 | Customer Service Clerk | SO | 1/1/2022 | \$36,834 | \$41,058 | \$45,766 | \$48,147 | \$50,532 | \$53,508 |
| C0671 | Clerk – Tax Adjustment | FCS | | | | | | | |
| C0675 | Administrative Assistant – Financial Planning & Policy | FCS | | | | | | | |
| C0678 | Security Assistant | EES | | | | | | | |
| C0680 | Bilingual Client Services Clerk | Planning | | | | | | | |
| C0681 | Administrative Assistant Technology Services | FCS | | | | | | | |
| C0682 | Data Processing Accounts Clerk | SO | | | | | | | |
| C0684 | Divisional Administrative Assistant | EES | | | | | | | |
| C0688 | Invoice and Processing Clerk | FCS | | | | | | | |
| C0689 | Administrative Assistant | Planning | | | | | | | |
| C0690 | Committee Support Clerk | FCS | | | | | | | |

| Class | Position Title | Department | Year | Start | 6 Months | 1 Year | 2 Years | 3 Years | 4 Years | |
|-------|--|------------|----------|----------|-------------|-----------|------------|------------|------------|------------|
| C0692 | Customer Service Representative - Licensing & By-laws | Planning | 1/1/2019 | \$34,846 | \$38,842 | \$43,296 | \$45,549 | \$47,804 | \$50,620 | |
| C0693 | Customer Service Representative – Permits | Planning | 1/1/2020 | \$35,456 | \$39,522 | \$44,054 | \$46,346 | \$48,641 | \$51,506 | |
| 96900 | Backflow Prevention & Monitoring Program Administrator | Planning | 1/1/2021 | \$36,112 | \$40,253 | \$44,869 | \$47,203 | \$49,541 | \$52,459 | |
| C0697 | Divisional Administrative Assistant | EES | 1/1/2022 | \$36,834 | \$41,058 | \$45,766 | \$48,147 | \$50,532 | \$53,508 | |
| C0698 | Financial Operations Assistant | FCS | | | | | | | | |
| 66900 | Customer Service Representative-City Clerk's Office | СМО | | | | | | | | |
| Code | Position Title | Department | Year | Start | 6 Months | 1 Year | 2 Years | 3 Years | 4 Years | 5 Years |
| C0722 | Supply Services Representative | FCS | 1/1/2019 | \$36,727 | \$40,606 | \$44,951 | \$46,957 | \$48,952 | \$50,952 | \$53,769 |
| C0738 | Administrative Coordinator- Recreation | SO | 1/1/2020 | \$37,370 | \$41,317 | \$45,738 | \$47,779 | \$49,809 | \$51,844 | \$54,710 |
| C0741 | Collections Assistant | CS | 1/1/2021 | \$38,061 | \$42,081 | \$46,584 | \$48,663 | \$50,730 | \$52,803 | \$55,722 |
| C0742 | Eligibility Review Support Officer | S | 1/1/2022 | \$38,822 | \$42,923 | \$47,516 | \$49,636 | \$51,745 | \$53,859 | \$56,837 |

| 000 | | | | | 0 | | c | c | - | - |
|-------|---|------------|-------------------|----------|----------|----------|----------|----------|----------|----------|
| Code | Position Title | Department | Year | Start | Months | Year | Years | Years | Years | Years |
| C0746 | Customer Service Representative – Revenue | FCS | 1/1/2019 | \$36,727 | \$40,606 | \$44,951 | \$46,957 | \$48,952 | \$50,952 | \$53,769 |
| C0747 | Financial Clerk | cs | 1/1/2020 | \$37,370 | \$41,317 | \$45,738 | \$47,779 | \$49,809 | \$51,844 | \$54,710 |
| C0748 | Industrial Land Assistant | FCS | 1/1/2021 | \$38,061 | \$42,081 | \$46,584 | \$48,663 | \$50,730 | \$52,803 | \$55,722 |
| C0758 | Administrative Assistant | СМО | 1/1/2022 \$38,822 | \$38,822 | \$42,923 | \$47,516 | \$49,636 | \$51,745 | \$53,859 | \$56,837 |
| C0760 | Documentation Services Representative | FCS | | | | | | | | |
| C0761 | Service Area Coordinator | SES | | | | | | | | |
| C0762 | Ontario Works Associate | CS | | | | | | | | |
| C0763 | Community Support Associate | SO | | | | | | | | |
| C0767 | Administrative Assistant – Realty Services | FCS | | | | | | | | |
| C0768 | Procurement Assistant | FCS | | | | | | | | |
| C0769 | Forestry Inspector | Planning | | | | | | | | |
| C0770 | Technician – Landfill Monitoring | EES | | | | | | | | |
| C0771 | Customer Service Representative - Parking | EES | | | | | | | | |
| C0773 | Laboratory Technician – Inside | EES | | | | | | | | |

| Class | Position Title | Department | Year | Start | 9 | ᠸ; | 7 | ლ ; | 4 | ر ئ |
|-------|--|------------|----------|----------|-------------|-----------|------------|------------|------------|------------|
| Code | | | | | Months | Year | Years | Years | Years | Years |
| C0774 | Administrative and Technical Support Representative - Planning | Planning | 1/1/2019 | \$36,727 | \$40,606 | \$44,951 | \$46,957 | \$48,952 | \$50,952 | \$53,769 |
| C0775 | Administrative and Technical Support Representative | Planning | 1/1/2020 | \$37,370 | \$41,317 | \$45,738 | \$47,779 | \$49,809 | \$51,844 | \$54,710 |
| C0776 | Housing Program Associate | HSSD | 1/1/2021 | \$38,061 | \$42,081 | \$46,584 | \$48,663 | \$50,730 | \$52,803 | \$55,722 |
| C0777 | Community Support Associate – Bilingual | HSSD | 1/1/2022 | \$38,822 | \$42,923 | \$47,516 | \$49,636 | \$51,745 | \$53,859 | \$56,837 |
| C0778 | Customer Service Representative | HSSD | | | | | | | | |
| C0779 | Facilities Operations Coordinator | FCS | | | | | | | | |
| Class | Position Title | Department | Year | Start | 6 Months | 1 Year | 2 Years | 3 Years | 4 Years | 5 Years |
| C0849 | Payroll Assistant | FCS | 1/1/2019 | \$40,841 | \$45,028 | \$49,644 | \$51,825 | \$54,000 | \$56,162 | \$56,924 |
| C0852 | Customer Relations Representative | EES | 1/1/2020 | \$41,556 | \$45,816 | \$50,513 | \$52,732 | \$54,945 | \$57,145 | \$57,920 |
| C0855 | Senior Cashier | FCS | 1/1/2021 | \$42,325 | \$46,664 | \$51,447 | \$53,708 | \$55,961 | \$58,202 | \$58,992 |
| C0856 | Collections Clerk | FCS | 1/1/2022 | \$43,172 | \$47,597 | \$52,476 | \$54,782 | \$57,080 | \$59,366 | \$60,172 |
| C0859 | Financial/Payroll Clerk | EES | | | | | | | | |

| Class | Position Title | Department | Year | Start | ဖ | τ- | 7 | ო | 4 | 2 |
|-------|---|------------|-------------------|----------|----------|----------|----------|----------|----------|----------|
| Code | | Coparation | - 0.0 | | Months | Year | Years | Years | Years | Years |
| C0860 | Accounting Clerk | FCS | 1/1/2019 | \$40,841 | \$45,028 | \$49,644 | \$51,825 | \$54,000 | \$56,162 | \$56,924 |
| C0861 | Inventory Control Clerk | FCS | 1/1/2020 \$41,556 | \$41,556 | \$45,816 | \$50,513 | \$52,732 | \$54,945 | \$57,145 | \$57,920 |
| C0862 | Budget Analyst | EES | 1/1/2021 | \$42,325 | \$46,664 | \$51,447 | \$53,708 | \$55,961 | \$58,202 | \$58,992 |
| C0863 | Mechanical Maintenance Person | EES | 1/1/2022 | \$43,172 | \$47,597 | \$52,476 | \$54,782 | \$57,080 | \$59,366 | \$60,172 |
| C0864 | Administrative Coordinator – Aquatics | SO | | | | | | | | |
| C0865 | Budget Analyst | FCS | | | | | | | | |
| C0867 | Tax Certificate and Apportionment Analyst | FCS | | | | | | | | |
| C0868 | Local Improvement & Assessment Analyst | FCS | | | | | | | | |
| C0869 | Energy Analyst | EES | | | | | | | | |
| C0870 | Accounts Receivable Clerk | FCS | | | | | | | | |
| C0872 | Inquiry Clerk | EES | | | | | | | | |
| C0874 | General Accounting Clerk | FCS | | | | | | | | |
| C0875 | Rent Supplement Coordinator | ПSSН | | | | | | | | |
| C0876 | Administrative Customer Service Representative | DCS | | | | | | | | |
| C0878 | Parking Services Technician | DCS | | | | | | | | |
| C0879 | Water and Wastewater Network Systems Associate | EES | | | | | | | | |

| ماموان | | | | | w | - | 0 | ۲ | V | ĸ |
|--------|---|------------|----------|----------|----------|----------|----------|----------|----------|----------|
| Code | Position Title | Department | Year | Start | Months | Year | Years | Years | Years | Years |
| C0948 | Waterworks Communications Coordinator | EES | 1/1/2019 | \$42,729 | \$47,130 | \$52,022 | \$54,292 | \$56,569 | \$58,835 | \$60,080 |
| C0951 | Senior Customer Service Representative – Revenue | FCS | 1/1/2020 | \$43,477 | \$47,955 | \$52,932 | \$55,242 | \$57,559 | \$59,865 | \$61,131 |
| C0955 | Senior Customer Service Clerk | SS | 1/1/2021 | \$44,281 | \$48,842 | \$53,911 | \$56,264 | \$58,624 | \$60,973 | \$62,262 |
| C0956 | Financial Clerk – POA | FCS | 1/1/2022 | \$45,167 | \$49,819 | \$54,989 | \$57,389 | \$59,796 | \$62,192 | \$63,507 |
| C0957 | Tax Accounting Clerk | FCS | | | | | | | | |
| 09600 | Housing Services Representative | Planning | | | | | | | | |
| C0964 | Accounts Payable Clerk | FCS | | | | | | | | |
| C0966 | Housing Coordinator | HSSD | | | | | | | | |
| C0967 | Laboratory Technician – Outside | EES | | | | | | | | |
| C0973 | Coordinator, Communications Tourism & Social Media | Tourism | | | | | | | | |
| C0974 | Senior Supply Services Representative | FCS | | | | | | | | |

| Class | Position Title | Department | Year | Start | 6 Months | 1 Year | 2 Years | 3 Years | 4 Years | 5 Years |
|-------|--|------------|----------|----------|-------------|-----------|------------|------------|------------|------------|
| C1029 | Employment Support Specialist | HSSD | 1/1/2019 | \$44,631 | \$49,318 | \$54,450 | \$56,854 | \$59,221 | \$61,658 | \$63,233 |
| C1039 | Committee Secretary | FCS | 1/1/2020 | \$45,412 | \$50,181 | \$55,403 | \$57,849 | \$60,257 | \$62,737 | \$64,340 |
| C1051 | Accounts Payable Supervisor | FCS | 1/1/2021 | \$46,252 | \$51,109 | \$56,428 | \$58,919 | \$61,372 | \$63,898 | \$65,530 |
| C1058 | Court Administration Clerk – POA | FCS | 1/1/2022 | \$47,177 | \$52,131 | \$57,557 | \$60,097 | \$62,599 | \$65,176 | \$66,841 |
| C1059 | GIS Data Technician | Planning | | | | | | | | |
| C1066 | Court Reporter/Clerk – POA | FCS | | | | | | | | |
| C1067 | Caseworker | cs | | | | | | | | |
| C1070 | Development Documentation Coordinator | DCS | | | | | | | | |
| C1072 | Integrated Land Planning Technologist | Planning | | | | | | | | |
| C1073 | Court Reporter/Clerk - POA Bilingual | FCS | | | | | | | | |
| C1076 | SDMT Caseworker | SO | | | | | | | | |
| C1078 | Fleet Services Analyst | EES | | | | | | | | |
| C1081 | Waitlist Coordinator | Planning | | | | | | | | |
| C1084 | Facilities Technologist 1 | FCS | | | | | | | | |
| C1085 | Records Systems Analyst | CMO | | | | | | | | |
| C1086 | Court Clerk | FCS | | | | | | | | |

| ō | | | | | c | , | C | c | • | L |
|-------|---|------------|----------|----------|-------------|-----------|------------|------------|------------|------------|
| Code | Position Title | Department | Year | Start | o Months | Year | 2 Years | 3 Years | 4 Years | 5 Years |
| C1087 | Procurement Specialist | FCS | 1/1/2019 | \$44,631 | \$49,318 | \$54,450 | \$56,854 | \$59,221 | \$61,658 | \$63,233 |
| C1088 | Financial Analyst | FCS | 1/1/2020 | \$45,412 | \$50,181 | \$55,403 | \$57,849 | \$60,257 | \$62,737 | \$64,340 |
| C1090 | Forestry Technologist | Planning | 1/1/2021 | \$46,252 | \$51,109 | \$56,428 | \$58,919 | \$61,372 | \$63,898 | \$65,530 |
| | | | 1/1/2022 | \$47,177 | \$52,131 | \$57,557 | \$60,097 | \$62,599 | \$65,176 | \$66,841 |
| Class | Position Title | Department | Year | Start | 6 Months | 1 Year | 2 Years | 3 Years | 4 Years | 5 Years |
| C1122 | Minor Variance Co-ordinator | Planning | 1/1/2019 | \$46,515 | \$54,318 | \$56,871 | \$59,397 | \$61,920 | \$64,433 | \$66,388 |
| C1140 | Planner I | Planning | 1/1/2020 | \$47,329 | \$55,269 | \$57,866 | \$60,436 | \$63,004 | \$65,561 | \$67,550 |
| C1141 | Special Events and Approvals Technologist | EES | 1/1/2021 | \$48,205 | \$56,291 | \$58,937 | \$61,554 | \$64,170 | \$66,774 | \$68,800 |
| C1143 | Laboratory Technologist I | EES | 1/1/2022 | \$49,169 | \$57,417 | \$60,116 | \$62,785 | \$65,453 | \$68,109 | \$70,176 |
| C1145 | Supervisor of Reception Licensing | FCS | | | | | | | | |
| C1146 | Current Budget Officer | FCS | | | | | | | | |
| C1150 | Accounts Payable & Processing Supervisor | SO | | | | | | | | |
| C1153 | Technician – Solid Waste Management | EES | | | | | | | | |
| C1155 | Research Analyst | Planning | | | | | | | | |
| C1161 | Animal Control & Welfare Coordinator | EES | | | | | | | | |

| Class | Position Title | Department | Vear | Start | 9 | - | 2 | ო | 4 | 2 |
|-------|---|------------|--------------|----------|-------------|--------------|------------|------------|------------|------------|
| Code | | Department | - מ <u>פ</u> | Olait | Months | Year | Years | Years | Years | Years |
| C1164 | Urban Design Technician | Planning | 1/1/2019 | \$46,515 | \$54,318 | \$56,871 | \$59,397 | \$61,920 | \$64,433 | \$66,388 |
| C1165 | Supervisor of Elections & Administrative Support | FCS | 1/1/2020 | \$47,329 | \$55,269 | \$57,866 | \$60,436 | \$63,004 | \$65,561 | \$67,550 |
| C1167 | Financial Analyst | CS | 1/1/2021 | \$48,205 | \$56,291 | \$58,937 | \$61,554 | \$64,170 | \$66,774 | \$68,800 |
| C1170 | Senior Tax Accounting Clerk | FCS | 1/1/2022 | \$49,169 | \$57,417 | \$60,116 | \$62,785 | \$65,453 | \$68,109 | \$70,176 |
| C1171 | Accounts Receivable Collections Officer | FCS | | | | | | | | |
| C1172 | Senior Accounting Clerk | FCS | | | | | | | | |
| C1173 | Infrastructure Systems Technologist | EES | | | | | | | | |
| C1174 | Asset Management System Analyst | FCS | | | | | | | | |
| C1175 | Coordinator, Asset Management (Regional Water Supply) | EES | | | | | | | | |
| Class | Position Title | Department | Year | Start | 6 Months | 1 Year | 2 Years | 3 Years | 4 Years | 5 Years |
| C1231 | Maintenance Management System Coordinator | EES | 1/1/2019 | \$48,422 | \$56,615 | \$59,305 | \$61,956 | \$64,612 | \$67,246 | \$69,541 |
| C1233 | Distribution Systems Coordinator | FCS | 1/1/2020 | \$49,269 | \$57,606 | \$60,343 | \$63,040 | \$65,743 | \$68,423 | \$70,758 |
| C1236 | Communications Specialist | CMO | 1/1/2021 | \$50,180 | \$58,672 | \$61,459 | \$64,206 | \$66,959 | \$69,689 | \$72,067 |
| C1237 | Technologist I | EES | 1/1/2022 | \$51,184 | \$59,845 | \$62,688 | \$65,490 | \$68,298 | \$71,083 | \$73,508 |

| Code | Position Title | Department | Year | Start | 6 Months | 1 Year | 2 Years | 3 Years | 4 Years | 5 Years |
|-------|---|------------|----------|----------|-------------|-----------|------------|------------|------------|------------|
| C1243 | Technologist I – Construction Administration | EES | 1/1/2019 | \$48,422 | \$56,615 | \$59,305 | \$61,956 | \$64,612 | \$67,246 | \$69,541 |
| C1246 | Financial Operations Analyst | CS | 1/1/2020 | \$49,269 | \$57,606 | \$60,343 | \$63,040 | \$65,743 | \$68,423 | \$70,758 |
| C1250 | Project Coordinator | Planning | 1/1/2021 | \$50,180 | \$58,672 | \$61,459 | \$64,206 | \$66,959 | \$69,689 | \$72,067 |
| C1251 | Community Services Support Specialist | S | 1/1/2022 | \$51,184 | \$59,845 | \$62,688 | \$65,490 | \$68,298 | \$71,083 | \$73,508 |
| C1252 | Supervisor – Tax Collections | FCS | | | | | | | | |
| C1254 | Compliance Coordinator | EES | | | | | | | | |
| C1255 | Security Services Technologist | EES | | | | | | | | |
| C1256 | Accounts Receivable & Collections Supervisor | SO | | | | | | | | |
| C1258 | Emergency Management Technologist | EES | | | | | | | | |
| C1260 | Architectural Plans Examiner I | Planning | | | | | | | | |
| C1261 | Environmental Outreach Coordinator | EES | | | | | | | | |
| C1262 | Fire Safety Technologist | EES | | | | | | | | |
| C1263 | Social Housing Technical Support Officer | Planning | | | | | | | | |
| C1264 | Children's Services Analyst | CS | | | | | | | | |

| Class | Position Title | Department | Year | Start | 6 Months | 1 Year | 2 Years | 3 Years | 4 Years | 5 Years |
|-------|--|------------|----------|----------|-------------|-----------|------------|------------|------------|------------|
| C1265 | Site Plan Approval Officer | Planning | | | | | | | | |
| C1268 | Data and Systems Analyst, Corporate Asset Management | | | | | | | | | |
| Class | Position Title | Department | Year | Start | 6 Months | 1 Year | 2 Years | 3 Years | 4 Years | 5 Years |
| C1325 | Payroll Supervisor | FCS | 1/1/2019 | \$50,245 | \$58,885 | \$61,675 | \$64,466 | \$67,285 | \$70,062 | \$72,692 |
| C1331 | Senior Laboratory Technologist | EES | 1/1/2020 | \$51,124 | \$59,915 | \$62,754 | \$65,594 | \$68,462 | \$71,288 | \$73,964 |
| C1338 | Facilities Systems Tradesperson | EES | 1/1/2021 | \$52,070 | \$61,023 | \$63,915 | \$66,807 | \$69,729 | \$72,607 | \$75,332 |
| C1341 | Capital Budget Officer | FCS | 1/1/2022 | \$53,111 | \$62,243 | \$65,193 | \$68,143 | \$71,124 | \$74,059 | \$76,839 |
| C1342 | Current Budget & Process Officer | FCS | | | | | | | | |
| C1344 | Procurement Officer | FCS | | | | | | | | |
| C1345 | Traffic Signal & Street Light Technologist | EES | | | | | | | | |
| C1347 | Housing Program Officer | Planning | | | | | | | | |
| C1349 | Assessment Review Officer | FCS | | | | | | | | |
| C1350 | Property Coordinator | FCS | | | | | | | | |
| C1351 | Industrial Land Officer | FCS | | | | | | | | |

| Class | Docition Tifle | Donortmont | Voor | Ctort | 9 | - | 2 | က | 4 | 2 |
|-------|--|------------|----------|----------|-------------|--------------|------------|------------|------------|------------|
| Code | | | 2 | סופור | Months | Year | Years | Years | Years | Years |
| C1352 | Plumbing Inspector | Planning | 1/1/2019 | \$50,245 | \$58,885 | \$61,675 | \$64,466 | \$67,285 | \$70,062 | \$72,692 |
| C1353 | Solid Waste Planning Coordinator | EES | 1/1/2020 | \$51,124 | \$59,915 | \$62,754 | \$65,594 | \$68,462 | \$71,288 | \$73,964 |
| C1356 | Water Demand Technologist | EES | 1/1/2021 | \$52,070 | \$61,023 | \$63,915 | \$66,807 | \$69,729 | \$72,607 | \$75,332 |
| C1358 | Transportation Technologist | EES | 1/1/2022 | \$53,111 | \$62,243 | \$65,193 | \$68,143 | \$71,124 | \$74,059 | \$76,839 |
| C1359 | Data & Reporting Analyst | NCFS | | | | | | | | |
| C1360 | Business and Zoning Coordinator | DCS | | | | | | | | |
| C1361 | Budget and Finance Analyst, Regional Water Supply | EES | | | | | | | | |
| C1362 | Municipal Law Enforcement Officer II | Planning | | | | | | | | |
| C1363 | Transportation Demand Management Coordinator | EES | | | | | | | | |
| Class | Position Title | Department | Year | Start | 6 Months | 1 Year | 2 Years | 3 Years | 4 Years | 5 Years |
| C1413 | Urban Economist | Planning | 1/1/2019 | \$52,197 | \$61,287 | \$64,210 | \$67,160 | \$70,062 | \$73,012 | \$75,844 |
| C1415 | Facilities Technologist – Interiors | EES | 1/1/2020 | \$53,110 | \$62,360 | \$65,334 | \$68,335 | \$71,288 | \$74,290 | \$77,171 |
| C1416 | Drainage Technologist | EES | 1/1/2021 | \$54,093 | \$63,514 | \$66,543 | \$69,599 | \$72,607 | \$75,664 | \$78,599 |
| C1417 | Technologist II | EES | 1/1/2022 | \$55,175 | \$64,784 | \$67,874 | \$70,991 | \$74,059 | \$77,177 | \$80,171 |

| Code | Position Title | Department | Year | Start | 6 Months | 1 Year | 2 Years | 3 Years | 4 Years | 5 Years |
|-------|---|------------|----------|----------|-------------|-----------|------------|------------|------------|------------|
| C1420 | Facilities Technologist – Architectural | EES | 1/1/2019 | \$52,197 | \$61,287 | \$64,210 | \$67,160 | \$70,062 | \$73,012 | \$75,844 |
| C1421 | Facilities Technologist – Mechanical | EES | 1/1/2020 | \$53,110 | \$62,360 | \$65,334 | \$68,335 | \$71,288 | \$74,290 | \$77,171 |
| C1422 | Technologist II | EES | 1/1/2021 | \$54,093 | \$63,514 | \$66,543 | \$69,599 | \$72,607 | \$75,664 | \$78,599 |
| C1424 | Technologist II | EES | 1/1/2022 | \$55,175 | \$64,784 | \$67,874 | \$70,991 | \$74,059 | \$77,177 | \$80,171 |
| C1425 | Business Systems Analyst | FCS | | | | | | | | |
| C1426 | Community Planning & Research Associate | NCFS | | | | | | | | |
| C1428 | Technologist II | EES | | | | | | | | |
| C1429 | Technologist II | EES | | | | | | | | |
| C1431 | Operations Technologist | EES | | | | | | | | |
| C1432 | Asset Management Systems Coordinator | EES | | | | | | | | |
| C1433 | Financial Services Supervisor & Systems Coordinator | FCS | | | | | | | | |
| C1434 | Property Appraiser/Negotiator | FCS | | | | | | | | |
| C1435 | Hardware Services Technician | FCS | | | | | | | | |
| C1437 | Asset Management Systems Coordinator | EES | | | | | | | | |
| C1438 | Financial Officer | Planning | | | | | | | | |

| Class | Position Title | Department | Year | Start | 6 Months | 1 Year | 2 Years | 3 Years | 4 Years | 5 Years |
|-------|--|------------|----------|----------|-------------|-----------|------------|------------|------------|------------|
| C1439 | Telecommunications Technologist | FCS | 1/1/2019 | \$52,197 | \$61,287 | \$64,210 | \$67,160 | \$70,062 | \$73,012 | \$75,844 |
| C1440 | Financial Systems Analyst | Planning | 1/1/2020 | \$53,110 | \$62,360 | \$65,334 | \$68,335 | \$71,288 | \$74,290 | \$77,171 |
| C1441 | Development Policy Coordinator | Planning | 1/1/2021 | \$54,093 | \$63,514 | \$66,543 | \$69,599 | \$72,607 | \$75,664 | \$78,599 |
| C1442 | Development Finance Technologist | Planning | 1/1/2022 | \$55,175 | \$64,784 | \$67,874 | \$70,991 | \$74,059 | \$77,177 | \$80,171 |
| C1443 | Urban Forestry Planner | Planning | | | | | | | | |
| C1447 | Parking Coordinator | SOG | | | | | | | | |
| C1448 | Planner II | Planning | | | | | | | | |
| C1449 | Geospatial Infrastructure Systems Coordinator | EES | | | | | | | | |
| C1450 | Development Inspection Technologist | SOG | | | | | | | | |
| C1451 | Site Development Planner | SOG | | | | | | | | |
| Code | Position Title | Department | Year | Start | 6 Months | 1 Year | 2 Years | 3 Years | 4 Years | 5 Years |
| C1503 | TSD Service Coordinator | FCS | 1/1/2019 | \$54,349 | \$63,835 | \$66,912 | \$69,995 | \$72,995 | \$76,073 | \$78,998 |
| C1510 | Heritage Planner | Planning | 1/1/2020 | \$55,300 | \$64,952 | \$68,083 | \$71,220 | \$74,272 | \$77,404 | \$80,380 |
| C1513 | Ecologist Planner | Planning | 1/1/2021 | \$56,323 | \$66,154 | \$69,343 | \$72,538 | \$75,646 | \$78,836 | \$81,867 |
| | | | | | | | | | | |

| Class Code | Position Title | Department | Year | Start | 6 Months | 1 Year | 2 Years | 3 Years | 4 Years | 5 Years |
|---------------|---|------------|----------|----------|-------------|-----------|------------|------------|------------|------------|
| C1514 | Water & Wastewater Control Systems Coordinator | EES | 1/1/2022 | \$57,449 | \$67,477 | \$70,730 | \$73,989 | \$77,159 | \$80,413 | \$83,504 |
| C1515 | Heritage & Museum Coordinator | Planning | 1/1/2019 | \$54,349 | \$63,835 | \$66,912 | \$69,995 | \$72,995 | \$76,073 | \$78,998 |
| C1517 | TSD Service Desk Supervisor | FCS | 1/1/2020 | \$55,300 | \$64,952 | \$68,083 | \$71,220 | \$74,272 | \$77,404 | \$80,380 |
| C1518 | Hardware Services Supervisor | FCS | 1/1/2021 | \$56,323 | \$66,154 | \$69,343 | \$72,538 | \$75,646 | \$78,836 | \$81,867 |
| C1519 | Survey Coordinator | EES | 1/1/2022 | \$57,449 | \$67,477 | \$70,730 | \$73,989 | \$77,159 | \$80,413 | \$83,504 |
| C1520 | Urban Designer | Planning | | | | | | | | |
| C1521 | Regional Water Control Systems Coordinator | EES | | | | | | | | |
| C1523 | Senior Technologist | EES | | | | | | | | |
| C1524 | Plan Building Inspector | Planning | | | | | | | | |
| Code | Position Title | Department | Year | Start | 6 Months | 1 Year | 2 Years | 3 Years | 4 Years | 5 Years |
| C1601 | Senior Technologist | EES | 1/1/2019 | \$56,520 | \$66,384 | \$69,583 | \$72,787 | \$75,910 | \$79,114 | \$82,149 |
| C1602 | Senior Technologist | EES | 1/1/2020 | \$57,509 | \$67,546 | \$70,801 | \$74,061 | \$77,238 | \$80,498 | \$83,587 |
| C1603 | Senior Technologist | SES | 1/1/2021 | \$58,573 | \$68,796 | \$72,111 | \$75,431 | \$78,667 | \$81,987 | \$85,133 |
| C1604 | Chemist | SES | 1/1/2022 | \$59,744 | \$70,172 | \$73,553 | \$76,940 | \$80,240 | \$83,627 | \$86,836 |

| Code | Position Title | Department | Year | Start | 6 Months | 1 Year | 2 Years | 3 Years | 4 Years | 5 Years |
|-------|---|------------|----------|----------|-------------|-----------|------------|------------|------------|------------|
| C1605 | Senior Transportation Technologist | EES | | | | | | | | |
| C1606 | Senior Facilities Technologist | EES | 1/1/2019 | \$56,520 | \$66,384 | \$69,583 | \$72,787 | \$75,910 | \$79,114 | \$82,149 |
| C1607 | Senior Property Appraiser Negotiator | FCS | 1/1/2020 | \$57,509 | \$67,546 | \$70,801 | \$74,061 | \$77,238 | \$80,498 | \$83,587 |
| C1609 | Senior Planner | Planning | 1/1/2021 | \$58,573 | \$68,796 | \$72,111 | \$75,431 | \$78,667 | \$81,987 | \$85,133 |
| C1610 | Landscape Architect | Planning | 1/1/2022 | \$59,744 | \$70,172 | \$73,553 | \$76,940 | \$80,240 | \$83,627 | \$86,836 |
| Class | Position Title | Department | Year | Start | 6 Months | 1 Year | 2 Years | 3 Years | 4 Years | 5 Years |
| C1700 | | | 1/1/2019 | \$58,690 | \$68,929 | \$72,254 | \$75,584 | \$78,822 | \$82,146 | \$85,303 |
| | | | 1/1/2020 | \$59,717 | \$70,135 | \$73,518 | \$76,907 | \$80,201 | \$83,584 | \$86,796 |
| | | | 1/1/2021 | \$60,822 | \$71,432 | \$74,878 | \$78,330 | \$81,685 | \$85,130 | \$88,402 |
| | | | 1/1/2022 | \$62,038 | \$72,861 | \$76,376 | \$79,897 | \$83,319 | \$86,833 | \$90,170 |

SCHEDULE B - CLOTHING AND SAFETY EQUIPMENT

The following are particulars of the Clothing and Safety Equipment and frequency of issue:

Environmental and Engineering Services Department and Planning and Development Staff or other staff as designated by the Corporation.

| Al | RTICLE | NUMBER | FREQUENCY |
|-------------------------|-------------------------|--------------|-------------------------------|
| Safety Shoes | | 1 | As required, no more |
| | | | frequently than 12 months |
| Safety Boots, Leat | her Bush or Thermal | 1 | As required, no more |
| | | | frequently than 12 months |
| The following will | be provided as require | d: | |
| Belts | Rain Suits | Hard Hat & L | iner |
| Knee Pads Gas Detectors | | Gloves | |
| Rubber Boots | Rubber Overshoes | Coveralls | |
| Miner's Lamp | Safety Glasses | Safety Vests | |
| Safety Goggles | (Prescription) | | |
| The cost of the ur | ndernoted item shall be | shared equal | ly by the Corporation and the |
| employee receivii | ng the item: | | |
| Parka or Three-in- | one Coat | · | As required |

Custodial Staff

| ARTICLE | NUMBER | FREQUENCY |
|-----------------------------|--------|---------------------------|
| Safety Shoes | 1 | As required, no more |
| | | frequently than 12 months |
| Shop Coats | 1 | As required |
| Wash & Wear | 3 | May 1 st |
| - Shirts | 3 | May 1 st |
| - Trousers or | 3 | May 1 st |
| - Ladies pantsuits | | |
| Uniform Purse | 1 | As required |
| Summer Issue | | |
| Jacket | 1 | As required |
| Blouses/Shirts | 3 | May 1 st |
| Skirts/Slacks | 2 | As required |
| Ties | 2 | As required |
| Raincoat/Rainsuit | 1 | As required |
| Spring & Fall Overcoat | 1 | As required |
| Hat | 1 | As required |
| Safety Shoes/Police Boots | 2 | As required, no more |
| | | frequently than 12 months |
| | | |
| | | |
| Winter Issue | | |
| Shirts/Turtle Neck Sweaters | 3 | Oct 1 st |

| Blouses/Turtle Neck Sweaters | 3 | Oct 1st |
|-------------------------------|---|---------------------------|
| Rubber Overshoes/Winter Boots | 1 | As required, no more |
| | | frequently than 12 months |
| Slacks | 2 | As required |
| Winter Hat | 1 | As required |
| Parkas/Three-in-one Coat | 1 | As required |
| Gloves (lined winter) | 1 | As required |
| Shop Coat | 1 | As required |

Parking Meter Staff

| ARTICLE | NUMBER | FREQUENCY |
|-------------------------------|--------|----------------------------------|
| Summer Issue | | |
| Jacket | 1 | As required, May 1 st |
| Blouses/Shirts | 3 | As required |
| Skirts/Slacks | 2 | As required |
| Ties | 2 | As required |
| Raincoat/Rainsuit | 1 | As required |
| Spring & Fall Overcoat | 1 | As required |
| Hat | 1 | As required, no more |
| | | frequently than 24 months |
| Safety Shoes/Police Boots | 2 | As required |
| Uniform - Purse | 1 | As required |
| Winter Issue | | |
| Shirts/Turtle Neck Sweaters | 3 | Oct 1 st |
| Blouses/Turtle Neck Sweaters | 3 | Oct 1st |
| Rubber Overshoes/Winter Boots | 1 | As required, no more |
| | | frequently than 12 months |
| Slacks | 2 | As required |
| Winter Hat | 1 | As required |
| Parkas/Three-in-one Coat | 1 | As required |
| Gloves (lined winter) | 1 | As required |
| Shop Coat | 1 | As required |

Environmental Services – Laboratory Staff

| ARTICLE | NUMBER | FREQUENCY |
|----------|--------|-----------|
| Shirts | 3 | Annually |
| Pants | 3 | Annually |
| Lab Coat | 1 | Annually |

Finance and Administration

| ARTICLE | NUMBER | FREQUENCY |
|----------------------------------|--------|------------------------------------|
| Stores Staff - Summer Issue | | |
| Safety Shoes | 1 | As required, no more |
| | | frequently than 12 months |
| Spring Jackets | 2 | May 1 st or as required |
| Shirts | 3 | As required |
| Slacks | 3 | As required |
| Winter Issue | | |
| Parka/Three-in-one Coat | 1 | As required |
| Raincoat and Hat | 1 | As required |
| (Prescription) Safety Glasses | 1 | As required |
| Safety Boots or Winter Overshoes | 1 | As required, no more |
| - | | frequently than 12 months |
| Shirts (long sleeve) | 3 | As required |
| Slacks | 3 | As required |

Procurement Officers

| ARTICLE | NUMBER | FREQUENCY |
|--------------|--------|-------------|
| Safety Shoes | 1 | As required |
| Safety Boots | 1 | As required |

Corporate Services

| ARTICLE | NUMBER | FREQUENCY | |
|--|--------|-------------|--|
| Customer Service Representative – Summer Issue | | | |
| Blouses/Shirts (short sleeve) | 3 | As required | |
| Skirts/Slacks | 3 | As required | |
| Blazer and/or Cardigan | 1 | As required | |
| Winter Issue | | | |
| | | | |
| Blouses/Shirts (long sleeve) | 3 | As required | |
| Skirts/Slacks | 3 | As required | |
| Jacket or Cardigan | 1 | As required | |

Employees who are actively working in the following positions and who conduct field inspection on a regular basis:

Plan Building Inspector

Plumbing Inspector

Municipal Law Enforcement Officer II

The above noted employees will wear the clothing issued by the Corporation when performing their job duties unless otherwise directed by their manager.

The clothing articles outlined will be issued in accordance with Article 19 of the Collective Agreement.

The employees described above will no longer receive a Clothing Committee allowance for jackets.

| ARTICLE | | NUMBER | FREQUENCY |
|---|----------------|--------------|-----------------------------------|
| Safety Shoes | | 1 | As required, no more |
| | | | frequently than 12 months |
| Safety Boots, Leather Bush or | Thermal | 1 | As required, no more |
| | | | frequently than 12 months |
| Pants | | 3 | May 1 st |
| Shirts | | 5 | May 1 st |
| Sweat Shirt or Fleece Zip Swe | ater | 1 | May 1 st |
| Parka or Three-in-one Coat | | 1 | October 1 st , no more |
| | | | frequently than 24 months |
| The Following will be provided as required: | | | |
| Belts | Rain Suits | | Gloves |
| Knee Pads | Gas Detectors | 3 | Coveralls |
| Rubber Boots Rubber Overshoes | | Safety Vests | |
| Miner's Lamp Safety Glasses | | | |
| Safety Googles | (Prescription) | | |
| | Hard Hat & Li | ner | |

Employees who are actively working in the following positions:

Hardware Services Supervisor

Hardware Services Technician

Telecommunications Group within the Business Systems Analyst classification

The above noted employees will be reimbursed for clothing articles up to the maximum amount as more specifically outlined below. Employees shall be reimbursed the amounts noted for clothing to be worn at work while performing their job duties. These amounts will be reimbursed following purchase and submission of proof purchase.

| ARTICLE | Amount | FREQUENCY |
|-------------------|------------|-------------------------------|
| Clothing for Work | \$230.00 | Annually per employee |
| | plus | |
| | applicable | |
| | taxes | |
| | | |
| 3 in 1 Jacket | \$160 plus | As required, no more |
| | applicable | frequently than once in every |
| | taxes | 24 months |

[&]quot;As required" in this schedule means replacement that is necessary as a result of normal wear and tear, and subject to applicable administrative procedures. Where frequency of issue is defined by date, this issue shall occur by that date or as soon as practical thereafter.

A uniform will be supplied to employees in the Customer Service Representative position and a uniform and protective footwear will be supplied to employees in the Records Clerk position, the specifications and frequency to be determined by Management.

Ratified on the 16 day of October 2018

CANADIAN UNION OF PUBLIC EMPLOYEES LOCAL UNION 101 London, Ontario

Steve Holland, President c/o Domenic Anello, Secretary C.U.P.E., Local101 1-380 Adelaide Street North London, Ontario N6B 3P6

Dear Mr. Holland:

The Corporation agrees that all rights, benefits and privileges which the employees now receive or possess shall continue to be enjoyed, possessed and held by the employees, as provided in the January 1, 2019 to December 31, 2022 Collective Agreement.

This letter shall form part of the Collective Agreement.

Ed Holder, Mayor

Catharine Saunders, City Clerk

BY HAND

ACKNOWLEDGED: (LOCAL 101)

Steve Holland, President

Domenic Anello, Secretary

APPENDIX A - RETIREES BENEFITS

For all employees with a hire date on or prior to July 22, 2015, a retiree is defined as either an employee who is retiring and who is eligible to receive a non-actuarially reduced OMERS pension or an employee who has at least twenty (20) (fifteen (15) for the term of the Collective Agreement), years of service and who is eligible to receive an actuarially reduced OMERS pension. For all employees with a hire date after July 22, 2015, a retiree is defined as an employee who is retiring and who is eligible to receive a non-actuarially reduced or actuarially reduced OMERS pension and who has at least thirty (30) years of service with the Corporation. In all cases, eligible to receive the pension means the pension will commence to be paid to the retiree effective the first day of the month following the month in which the employee retires.

The carrier of the retirees group will be established in conjunction with the Corporation employee group carrier.

The specific criteria for eligibility are:

- The retiree may elect single or dependent coverage as applicable.
- Eligibility to participate in the Plan ceases at age sixty-five (65) for any person insured.
- A spouse of a deceased retiree may continue participation until the earlier of the:
 - (i) Date the retiree would have attained age sixty-five (65) years, or;
 - (ii) The date the surviving spouse remarries, or;
 - (iii) The date the survivor attains the age of sixty-five (65).
- The retiree must enrol in the Plan at the time of retiring and must continue uninterrupted participation in order to be eligible to participate in the Plan.
- The retiree must be domiciled in Canada.
- The retiree will waive any right to claim against the Corporation and the Carrier in the event coverage lapses by reason of any act or omission on the retiree's part in fulfilling any of the terms and conditions of the plan.

APPENDIX B - LONG TERM DISABILITY DETAILS

ELIGIBILITY All full time active employees. New employees upon

completion of probationary period.

COMPULSORY BENEFIT Yes.

WAITING PERIOD Minimum of one hundred ninety-six (196) calendar days

but not before exhaustion of sick leave and Short-term

disability benefits.

BENEFIT LEVEL Seventy percent (70%) of salary at the time of disability.

MAXIMUM BENEFIT Three thousand dollars (\$3,000.00) per month.

BENEFIT DURATION The earlier of recovery, age sixty-five (65) or death.

OFFSETS Employee Canada Pension Benefit (not dependent)

and Workplace Safety and Insurance Act if seventy percent (70%) benefit level. If eighty-five percent (85%) all income sources participated in by the employer and employee, e.g. OMERS disability, Canada Pension disability benefit and Workplace Safety and Insurance

Board benefits if applicable.

PRF-EXISTING CONDITIONS This clause is waived.

OWN OCCUPATION Two (2) years own occupation. Beyond that, any

occupation. Also the L.T.D. Plan provides for a two (2) year rehabilitation period with a fifty percent (50%)

benefit level.

NERVOUS AND MENTAL

RESTRICTION

Must be under the care of a psychiatrist. If not, benefit

could be reduced to a fifty percent (50%) level.

WAIVER OF PREMIUM

No premium is payable if eligible for benefit.

UNDERWRITING METHOD Fully pooled as opposed to experience rating.

PREMIUM SHARING One hundred percent (100%) paid by the employer.

IMPLEMENTATION DATE October 1, 1985; revised January 1, 1992 and June 1,

2003

| • | LEAVES OF ABSENCE |
|---|---|
| • | ONTARIO WORKS PROGRAM |
| • | REDUCED HOURS OF OPERATION (HOLIDAY CLOSURE) |
| • | TESTING |
| • | FLEXIBLE WORK SCHEDULE PROGRAM AND COMPENSATORY TIME GUIDELINES |
| • | CONVENTION SERVICING |
| • | TERMS OF DEDICATED PRESIDENCY FOR LOCAL 101 |
| • | MUNICIPAL LAW ENFORCEMENT OFFICER II |
| • | JOB SHARING |
| • | CALL IN & STAND-BY - REMOTE ACCESS |
| • | EXCESS HOURS OF WORK AGREEMENT |
| • | UNITED WAY SPONSORED EMPLOYEE PROGRAM |
| • | HIGH SCHOOL CO-OP PLACEMENT |
| | |

All letters of understanding listed append to and become part of the Collective Agreement.

SEASONAL TOURISM COUNSELLOR/PAID LUNCH UNDER

EMPLOYEES HIRED ON OR BEFORE JULY 22, 2015 - JOB

REFERRAL TO ARBITRATION OF A JOB EVALUATION MATTER

ARTICLE 10

POSTING AND "BUMPING"

I FTTER OF LINDERSTANDING

BETWEEN: CANADIAN UNION OF PUBLIC EMPLOYEES, LOCAL 101 (hereinafter called Local 101)

AND

THE CORPORATION OF THE CITY OF LONDON (hereinafter called The City or The Corporation)

Re: Leaves of Absence

Whereas Local 101 has filed grievance Policy 7-96 relating to the granting of leaves of absence;

And whereas the Parties wish to resolve this dispute without the need for litigation;

They hereby agree as follows:

- This agreement relates solely to the granting of leaves of absence for personal reasons, which are not emergencies. Emergency leaves of absence shall still be governed by Article 21.2 and the remainder of the Collective Agreement.
- 2. Notwithstanding any provision of the Collective Agreement between Local 101 and the City which restricts the granting of leaves of absence, an employee who desires a leave of absence without pay for personal reasons shall make a request in writing to the applicable Managing Director, as far in advance as possible, setting out the time period for which the leave is requested.
- 3. Having regard to all the circumstances surrounding the request including its operational requirements, the Corporation may, pursuant to the Letter of Understanding, in its discretion grant an employee a leave of absence without pay for personal reasons in accordance with this agreement.
- 4. The City shall not grant personal leaves of absence exceeding twelve (12) consecutive months. In addition, the City shall not grant any employee more than eight (8) personal leaves which last one (1) day or more and less than five (5) days per calendar year.
- 5. Notwithstanding the provisions of the Collective Agreement, group insurance benefits, seniority service accumulation, the payment of holiday pay, and the earnings of vacation credits shall cease thirty (30) days after the commencement of the leave of absence and will not resume until the employee returns to active employment, except as provided below.

- Where a leave of absence without pay is granted to an employee for military/reserve service, the employee's seniority, service and vacation credits will continue to accumulate.
- 7. Employees who take a personal leave of absence which is in excess of thirty (30) consecutive calendar days, and who wish to continue group health and dental benefits may do so at their own cost by requesting in writing that all or certain benefits be continued and making payment of the applicable premiums to the City in advance.
- 8. The granting of leaves of absence pursuant to this Letter of Understanding shall not interfere with the scheduling of planned vacation pursuant to Article 12.6.
- 9. This agreement shall have no effect or application on existing arrangements for union leaves of absence, education allowances, bereavement leave, pregnancy and parental leave, jury duty or citizenship leave.

This agreement was ratified on the 16th of October 2018, on behalf of the Parties.

FOR THE UNION

Steve/Hølland,

President, C.U.P.E. Local 101

FOR THE CORPORATION

William C. Coxhead

BETWEEN:

CANADIAN UNION OF PUBLIC EMPLOYEES, LOCAL 101 (hereinafter called Local 101)

AND

THE CORPORATION OF THE CITY OF LONDON (hereinafter called The City or The Corporation)

Re: Ontario Works Program

The parties hereby agree as follows, on the understanding that "employees" refers to employees governed by the Local 101 Collective Agreement.

- This agreement relates solely to the impact of the Ontario Works program relative to Article 18 of the Collective Agreement, or to other similar programs which may be instituted by the Local, Provincial or Federal government.
- 2. Notwithstanding the scope of Article 18 of the Collective Agreement, both parties agree that:
 - No Ontario Works program placements will be made outside of the provisions of the Collective Agreement.
 - Placements will not violate any Collective Agreement provision governing the assignment of work.
 - Placements must not displace any paid employment within the bargaining unit. Examples:
 - position currently held by an employee;
 - position of an employee who has been laid off and has recall rights under a collective agreement;
 - position of an employee who is on a leave of absence;
 - a collective of duties previously held by employees, within a minimum of three years.

This agreement was ratified on the 16th of October 2018, on behalf of the Parties.

FOR THE UNION

Steve Molland,

President, C.U.P.E. Local 101

FOR THE CORPORATION

William C. Coxhead

BETWEEN:

CANADIAN UNION OF PUBLIC EMPLOYEES, LOCAL 101 (hereinafter called The Union)

AND

THE CORPORATION OF THE CITY OF LONDON (hereinafter called The Employer)

Re: Reduced Hours of Operation (Holiday Closure)

City Hall and related facilities applicable to CUPE Local 101

The Employer and The Union agree to the following with respect to reduction of services and related facilities and/or operations between Boxing Day and noon of New Year's Eve for the term of the January 1, 2019 to December 31, 2022 Collective Agreement (known as Holiday Closure Period):

- Council may, in its sole discretion, decide on a closure/reduction of hours during any or all of the Holiday Closure Periods.
- In the event of such declaration, all operations except those deemed essential by the employer will be halted.
- 3. Employees may use time owing in the form of vacation time or accumulated time in lieu of overtime, compensatory time to continue pay during this period.
- 4. Except as noted in paragraph 3 above, all employees not required to provide essential services shall be considered to be temporarily laid off for the period of closure, and shall not receive pay for the period of closure.
- 5. It is specifically agreed that employees who would otherwise be entitled to sick disability payments during any such period except as per Article 12.4 will not be entitled in the event of a declaration of closure, but that employees who would otherwise be entitled to STD, LTD or WSIB related benefits will remain entitled during any periods of closure.
- Grievances or other actions taken by the union with respect to Holiday Closure Periods shall be limited to those dealing with the application of this agreement.

7. This agreement is made without precedent or prejudice, particularly with respect to either Party's rights at the conclusion of the agreement.

This agreement was ratified on the 16th of October 2018, on behalf of the Parties.

FOR THE UNION:

FOR THE CORRORATION:

Steve/Halland

President, C.U.P.E. Local 101

William C. Coxhead

BETWEEN:

THE CORPORATION OF THE CITY OF LONDON (hereinafter called the Corporation)

AND

CANADIAN UNION OF PUBLIC EMPLOYEES - LOCAL 101 (hereinafter called the Union)

Re: Testing

Whereas the Union and the Corporation wish to mutually address the issue of testing during job competitions in order to reduce or eliminate any future disputes on the issue, and to provide for policies and procedures for such testing all in accordance with the Collective Agreement, they mutually agree as follows:

- The Union recognizes the Corporation's right to test employees in the context of a
 job competition, in accordance with this Letter of Understanding.
- If the Corporation wishes to conduct a test, it shall indicate on the job posting that applicants may be required to take a test.
- Applicants who have the necessary qualifications will be given reasonable
 advance written notice of the date of the test, which notice shall be given at least
 nine (9) working days in advance of the test in all cases with the exception of
 typing tests.
- 4. Where a written test is conducted the Corporation shall make readily available to any employee who is to take the test, written material from which to study for the test. The written test shall be based on this written material which may include books, drawings, formulae, tables, documents, manuals, plans, policies, statutes, regulations, standards, or any other written material provided in a text or electronic format. If the City provides material by electronic format it shall also have the material readily available in text format or hard copy. Any applicant may request the material in text format or hard copy. Employees shall be notified where this material can be obtained on the notice provided for in paragraph three (3). This material shall be provided without charge to the employee.
- 5. The notice of the test shall also describe the general subject matter of the test.
- 6. In any case, all tests, and the administration of the testing process, including marking, will be:
 - (a) Fair, unbiased and reasonable; and

- (b) Designed to, and actually, assist the Corporation in determining whether the employee has the necessary skill and ability to perform the duties involved.
- 7. The Corporation may set a threshold for performance on a test (e.g. typing test), or a passing mark for the test, which passing mark shall not exceed sixty percent (60%). However, the Corporation shall not rely exclusively on a test result in determining whether an employee has the necessary skill and ability to perform the duties involved in the posted position, unless the test reveals conclusively that the employee does not have the necessary skill and ability to perform the duties involved in the posted position.
- Employees who successfully complete a test, but are not appointed to the posted job, will not need to be retested for the same job classification, for a period of one year.
- 9. This agreement does not alter, or amend, or in any way affect the parties rights under the Collective Agreement, in particular Article 8.
- It is understood that this Letter of Understanding applies only to C.U.P.E. Local 101.

This agreement was ratified on the 16th of October 2018, on behalf of the Parties.

FOR THE UNION

Steve Holland.

President, C.U.P.E. Local 101

FOR THE CORPORATION

William C. Coxhead

BETWEEN:

CANADIAN UNION OF PUBLIC EMPLOYEES, LOCAL 101 (hereinafter called Local 101)

AND

THE CORPORATION OF THE CITY OF LONDON (hereinafter called The City or The Corporation)

Re: Flexible Work Schedule Program and Compensatory Time Guidelines

- Adequate staffing must be maintained in all work areas in order to maintain the highest possible level of public service. Flexible Work Schedule Program ("The Program") and Compensatory Time may not be available in all work areas.
- 2. Participation is voluntary for all employees.
- 3. A complete normal work day and work week will consist of seven (7) hours and thirty-five (35) hours respectively.
- 4. In all cases start and quit times will be on the hour or half hour to facilitate timekeeping.
- 5. All employees must take a lunch break of not less than thirty (30) minutes nor more than two (2) hours. Lunch breaks must be in thirty (30) minute increments. An employee who wishes to take a thirty (30) minute lunch break for the purpose of shortening a work day or accumulating compensatory time must obtain prior management approval. The Program and/or Compensatory Time does not allow an employee to work continuously with no lunch break or rest periods in order to shorten the work day or to accumulate compensatory time.
- 6. Provisions to the Collective Agreement concerning hours of work, overtime, and meal allowances will be set aside and not apply to The Program and Compensatory Time. However, it is important to note that The Program and Compensatory Time does not preclude authorized overtime as defined in the Collective Agreement.
- The Flexible Work Schedule Program and Compensatory Time Guidelines and the application of each may be amended from time to time through mutual agreement of the Parties

8. Management approval is required for any participation by an employee in The Program. Any specific work schedule within the guidelines of the Program must be approved by the appropriate manager in advance. Management in all Departments will fairly and reasonably consider all requests and no request will be denied except as provided for in Point 1.

Management reserves the right to withdraw approval of a flex time schedule if the guidelines or work schedule are not adhered to or there is a negative public service impact.

 The Corporation may establish normal core periods that may vary by work area during which any employee scheduled to work must be at work except for the lunch break

The normal bandwidth period is 7:00 a.m. to 8 p.m.

- Normally an employee must establish a work schedule on a bi-weekly or longer basis and obtain prior managerial approval. Amendments to the schedule must be approved in the same manner.
- Regardless of any pre-approved work schedule an authorized absence such as sick leave credits, vacation, jury duty, bereavement leave, leave for Union business, etc., will be equal to seven (7) hours.
- Should Management require an employee to work outside the hours of preapproved flexible work schedule it will be considered overtime as defined in the Collective Agreement.

Compensatory Time Guidelines

13. Equal and compensatory time off may be accumulated to a maximum of thirty-five (35) hours at any time. Normally compensatory time may be taken to a maximum of seven (7) hours at a time in any two (2) consecutive weeks. Compensatory time in excess of seven (7) hours at a time, may be permitted with managerial approval. Such time off requires managerial approval subject to the staffing requirements. Also, compensatory time off will be scheduled equitably amongst employees.

Should an employee have five (5) days accumulated the employee must use some compensatory time before accumulating any further time.

- Normally an employee may work no more than nine (9) hours in a day or forty (40) hours in a week.
- 15. Hours worked in excess of seven (7) per day and equal compensatory time off will be recorded utilizing the current payroll system.

16. No employee will receive cash in lieu of compensatory time.

This agreement was ratified on the 16th of October 2018, on behalf of the Parties.

FOR THE UNION

FOR THE CORPORATION

Steve/Holland,

President, C.U.P.E. Local 101

William C. Coxhead

BETWEEN:

THE CORPORATION OF THE CITY OF LONDON (hereinafter called the Corporation)

AND

CANADIAN UNION OF PUBLIC EMPLOYEES - LOCAL 101 (hereinafter called the Union)

Re: Convention Servicing

The Parties undersigned hereby agree to the following terms which take precedence over, and replace any conflicting provisions in the Collective Agreement between the parties. These terms apply to the position of Convention Servicing and Sales Representative of Tourism London Inc., and the Program Coordinators (Recreation) in Community Services.

- The hours of work for these positions shall be completely defined by this Letter of Understanding. A regular work period shall be seventy (70) hours scheduled over fourteen (14) calendar days which will coincide with the pay period. A regular day shall be up to and including (10) hours in length. The Employer will endeavor to schedule successive days off in the work period.
- 2. The manager may schedule work days in excess of ten (10) hours. Any hours scheduled and worked in excess of ten (10) hours per day shall be deemed to be "excess hours". Excess hours are to be credited to the incumbent at time and a half and a current record shall be maintained. Excess hours will be taken as paid time off at times which is mutually acceptable to the incumbent and manager. In no circumstances will overtime premium be paid for excess hours.
- The premium is not applicable to these positions, except that weekend shift premium in accordance with the current rate negotiated in Article 10.4 (b) of the Collective Agreement, shall be paid for all hours scheduled and worked between midnight Friday and midnight Sunday.
- 4. In the event of a job posting for the subject positions, the posting will draw to the attention of applicants the fact that this Letter of Understanding is in force.
- 5. All other terms of the Collective Agreement will prevail.

This agreement is signed without Prejudice or Precedent to either party.

This agreement was ratified on the 16th of October 2018, on behalf of the Parties.

Steve/Holland, President, C.U.P.E. Local 101

FOR THE CORPORATION

William C. Coxhead

BETWEEN:

THE CORPORATION OF THE CITY OF LONDON (The Employer)

AND

CANADIAN UNION OF PUBLIC EMPLOYEES, LOCAL 101 (The Union)

Re: Terms of Dedicated Presidency for Local 101

The following terms apply for the duration of the Collective Agreement and are renewable by mutual agreement.

- The President of the Local shall be deemed to be dedicated (i.e., full-time) in the capacity as President, and shall receive pay for 1827 hours per annum inclusive of all vacation, bereavement, statutory holidays and any other paid leave. In the event of strike or lockout, pay shall cease for the period of strike or lockout.
- The President shall be paid at the salary rate of their position per Schedule A of the Collective Agreement.
- 3. The employer may backfill for the position held by the President on a temporary basis, while the individual is appointed as President, and fill any vacancy so created on the same temporary basis. If an employee elected as President is not re-elected to the office, or for any other reason ceases to be President, that employee shall be returned to the position formerly held at the conclusion of the term of office, as shall any employee appointed to replace on a temporary basis in accordance with the Collective Agreement.
- 4. The President may apply for any posted vacancies. If the President is the successful applicant his/her salary rate and level will reflect the new level as per Schedule A of the Collective Agreement effective the date of appointment to the new position.
 - In the event of a layoff, should the position held by the President be displaced, the President will be entitled to "bump" in accordance with Article 28.3 or may apply and be appointed to one (1) vacancy and his/her salary rate and level will reflect the new level as per Schedule A of the Collective Agreement effective the date of the appointment to the new position.
- 5. All insured benefits, sick leave and seniority shall continue as if the employee appointed as President had continued in the position per section 3 above.

Further, the employee will not be entitled to overtime, standby or meal allowance.

6. The President shall continue to be deemed to be an employee of the Corporation and at work for all purposes, including WSIB. Any sickness or injury shall be reported immediately to the delegated Manager, Human Resources. It is specifically noted that the Parties agree that the Union is deemed to be the employer for all purposes, including WSIB when the President is engaged in any activities not related to the negotiation or administration of the Collective Agreement between the City of London and CUPE. Local 101.

This agreement was ratified on the 16th of October 2018, on behalf of the Parties.

FOR THE UNION

Steve Halland,

President, C.U.P.E. Local 101

FOR THE CORPORATION

William C. Coxhead

BETWEEN:

THE CORPORATION OF THE CITY OF LONDON (The Employer)

AND

CANADIAN UNION OF PUBLIC EMPLOYEES, LOCAL 101 (The Union)

Re: Municipal Law Enforcement Officer II

- 1. The parties acknowledge that the public and the Council of the City of London have requested weekend and evening enforcement of municipal by-laws.
- Therefore the parties agree to change the hours of work for the job Municipal Law Enforcement Officer II. This position will be in addition to the existing complement of Municipal Law Enforcement Officers, unless the Union is notified otherwise.
- The incumbents in the current Municipal Law Enforcement Officer job will not be required to work other than normal hours, but may be required to work overtime as assigned in accordance with Article 10.6.
- The following shift definition will apply only to the Municipal Law Enforcement Officer II job.
- 5. A shift shall mean any seven (7) hour work day, plus one (1) hour off for lunch for any consecutive five (5) day period other than the normal work day as set out in Article 10.1 of the Agreement. Shifts will be determined by management and will start no earlier than 8:30 a.m. and end no later than 3:00 a.m. The Municipal Law Enforcement Officer II would be required to work evenings and some weekends. Fourteen (14) days' notice would be given for setting the schedule in consultation with the employee. Schedules will cover, at least a four (4) week period.
- Shift premium will be paid in accordance with Article 10.4 of the Collective Agreement.
 When required to work overtime or on a holiday, pay will be in accordance with Article 10.6(a) of the Collective Agreement.

This agreement was ratified on the 16th of October 2018, on behalf of the Parties.

FOR THE UNION

Steve Holland

President, C.U.P.E. Local 101

FOR THE CORRORATION

William C. Coxhead

BETWEEN:

THE CORPORATION OF THE CITY OF LONDON (The Employer)

AND

CANADIAN UNION OF PUBLIC EMPLOYEES, LOCAL 101 (The Union)

Re: Job Sharing

Job Sharing is an arrangement which allows two qualified employees to equally divide the hours, responsibilities, wages, and benefits of one full-time position according to the provisions of an agreement between the job share partners, the Corporation, and the Union. As such, the shared position continues to be identified as a full time position. Job share proposals will be considered for approval by the Employer, and the Union provided there is no adverse impact on service delivery and no net increase in cost to the Employer, and that the following Guidelines are adhered to:

Program Participation and Eligibility:

Proposals for job sharing must be employee driven. Only employees in full-time permanent positions, who are fully qualified for the shared position, may initiate a job sharing proposal. Qualifications will be assessed by the Corporation and may involve testing.

- 1. Casual, part-time, contract, and temporary positions are exempt from this program. Eligible Job Share Employees must not be in probationary or trial periods or temporary assignments (outside their "home" position).
- 2. The hours of work for the two employees will be divided such that each employee will work fifty percent (50%) of the hours of the full time position.
- 3. A proposal to job share will be submitted by the employees to their respective Managers. The Manager for the position to be job shared will consider the proposal for approval, having regard for the employees' qualifications and any adverse impact on service levels. In the event that a proposal is refused, the Manager will provide reasons, in writing, to the employees.
- 4. When a job sharing proposal is supported, employees will be required to enter in a job sharing agreement. Approval of a job sharing arrangement will require the signature of the employees involved the Union, and the Corporation.

- 5. One (1) of the job share employees will relinquish the rights to his/her position and that position will be posted as a full-time permanent vacancy at such time as the job sharing agreement is signed.
- 6. When an employee leaves a shared position, the remaining employee will revert back to full-time hours, with fourteen (14) days advance notice.
- 7. Job share employees retain the right to apply for job vacancies in accordance with the terms of the Collective Agreement.
- 8. Seniority, benefits, sick leave, public holidays and vacation will be provided in the same manner as part-time employees.
- 9. Employees entering into job sharing arrangements shall have satisfactory job performance and attendance.
- 10. Where employees interested in job sharing are in different classification levels, the shared position will be the one at the lower classification with salary adjustments in accordance with Article 9.4 (b).
- 11. A job share employee shall not be required to cover the absence of a job share partner in the event of illness, (other than an extended period of short term sick leave, or long term disability), vacation, or bereavement unless agreed to by all parties to the job sharing agreement. Any additional work agreed to by the Manager and the employee would be at straight time rates up to thirty-five (35) hours per week.
- 12. If the Corporation finds the job-sharing arrangement is not satisfactory during the first ninety (90) days, it shall give fourteen (14) days written notice of the termination of the job share arrangement to the job share partners and the Union. All employees affected by the job share will revert to their former position per Article 8.3 (a) of this Collective Agreement.
- 13. Job sharing arrangements shall not exceed five percent (5%) of the current full-time regular complement of employees in any given Department unless otherwise agreed to by both Parties.
- 14. This Letter of Understanding will be in place until the terms of a new Collective Agreement has been negotiated.

This agreement was ratified on the 16th of October 2018, on behalf of the Parties.

FOR THE UNION

Steve Holland,

President, C.U.P.E. Local 101

FOR THE CORPORATION

William C. Coxhead

BETWEEN:

THE CANADIAN UNION OF PUBLIC EMPLOYEES AND ITS LOCAL 101 (the "Union")

AND

THE CORPORATION OF THE CITY OF LONDON (the "Corporation")

Re: Call In & Stand-by - Remote Access

The parties agree as follows:

- Technological support traditionally performed by the Business Systems Analysts during the "off hours" will be provided by the Business Systems Analysts or other qualified members of Local 101.
- The compensation for the work performed in point 1 will be compensated as follows:
 - (i) Employees must be on stand-by in accordance with Article 10.12
 - (ii) Work performed by remote access solely will be compensated at the appropriate overtime rate for all work performed to the nearest fifteen (15) minutes (1/4 hour)
 - (iii) Work requiring a visit to the actual work location will be in accordance with Article 10.9
- The above procedure may be extended to other Union employees as appropriate and as requested by the Corporation.

This agreement was ratified on the 16th of October 2018, on behalf of the Parties.

FOR THE UNION

Steve Holland.

President, C.U.P.E. Local 101

FOR THE CORPORATION

William C. Coxhead

BETWEEN

THE CANADIAN UNION OF PUBLIC EMPLOYEES AND ITS LOCAL 101 (the "Union")

AND

THE CORPORATION OF THE CITY OF LONDON (the "Corporation")

Re: Excess Hours of Work Agreement

WHEREAS the <u>Employment Standards Act, 2000</u> requires that the parties have a written agreement regarding hours worked in excess of forty-eight (48) hours per week:

NOW THEREFORE the parties agree as follows:

- The parties agree that Union members may work beyond their regularly scheduled workday but not to exceed thirteen (13) hours in a workday.
- 2. The parties agree that Union members working in the following classifications may be scheduled/requested to work by the Corporation hours in excess of forty-eight (48) hours and up to sixty (60) hours per week:
 - Municipal Election Clerk
 - Administrative Assistant-Financial Planning and Policy
 - Treasury Management Officer
 - Payroll Systems Coordinator
 - Business Systems Analyst
 - Hardware Services Technician
 - Hardware Services Supervisor
 - Water and Wastewater Control Systems Coordinator
 - Technologist II
 - Senior Technologist II
 - Budget Clerk
 - Budget Analyst
 - Capital Budget Officer
 - · Current Budget and Process Officer
 - Committee Secretary
 - · Senior Accounting Clerk
 - Financial Services Supervisor
 - Financial Systems Analyst
 - Inventory Control Clerk

- Senior Supply Services Representative
- Supply Services Representative
- 3. The parties acknowledge and agree that scheduling or requests by the Corporation to work hours in excess of forty-eight (48) hours and up to sixty (60) hours per week will be in accordance with the Collective Agreement between the Union and the Corporation as well as any applicable legislation. Should the provision(s) of the Collective Agreement and this Agreement conflict; the provisions of the Collective Agreement shall prevail.
- The parties agree that this Agreement will be effective upon execution, but its implementation will be subject to the approval of the Director, Employment Standards or as otherwise provided in the Employment Standards Act. 2000.
- 5. Either party may revoke this Agreement upon the provision of ninety (90) calendar day's written notice.

This agreement was ratified on the 16th of October 2018, on behalf of the Parties.

FOR TH**E** UNION

Stevelle Mayata

President C.U.P.E. Local 101

FOR THE CORPORATION

William C. Coxhead

BETWEEN

THE CORPORATION OF THE CITY OF LONDON (the "Corporation")

AND

THE CANADIAN UNION OF PUBLIC EMPLOYEES AND ITS LOCAL 101 (the "Union")

Re: United Way Sponsored Employee Program

Whereas it is the desire of the parties to participate in the above program which is open to all Corporation employees to apply, and

Whereas this program provides for the successful applicant to participate as a sponsored full time employee to the United Way ("Sponsored Employee") for a temporary period of time spanning no longer than four (4) months;

And whereas the parties wish to clarify some employment terms for Union member(s) should they participate as a Sponsored Employee;

They do hereby agree as follows:

- The Sponsored Employee position will be considered a temporary position outside of the bargaining unit.
- Notwithstanding Article 6.6 of the Collective Agreement, the Union member who participates as a Sponsored Employee will continue to earn seniority.
- 3. Article 10 of the Collective Agreement, the Flexible Work Schedule Program Guidelines Letter of Understanding and Appendix Agreement 1 Article 13 and 14 will not apply to the Union member while participating as a Sponsored Employee. It is understood that the Union member will be compensated based on her/his current weekly earnings for a maximum of thirty-five (35) hours per week (forty (40) hours per week for former PUC Employees employed in a former PUC position).

4. Except as amended by this Agreement, all provisions of the Collective Agreement between the parties will apply.

This agreement was ratified on the 16th of October 2018, on behalf of the Parties.

FOR THE UNION

Steve Holland,

President, C.U.P.E. Local 101

FOR THE CORPORATION

William C. Coxhead

BETWEEN:

THE CORPORATION OF THE CITY OF LONDON (the "Corporation")

AND

THE CANADIAN UNION OF PUBLIC EMPLOYEES AND ITS LOCAL 101 (the "Union")

Re: High School Co-op Placement

The Union and the Corporation of the City of London agree to the participation in the High School Co-operative Education program from the time period between September and June. The selection of the co-op placement and the location will be at the discretion of the Corporation of the City of London and be managed and supervised by a non-union employee. It is understood by all parties that high school co-op positions are an unpaid placement.

In the event the Corporation decides to reduce the numbers of the permanent staff complement, or, in the event the Corporation decides to not fill a vacancy or vacancies occurring in the permanent staff complement, it is agreed that such vacancy should not be filled by the hiring of a co-op student or other student.

It is understood the co-op students are supernumerary and their utilization will not result in lay-off of any employee covered by this agreement nor will co-op students be used to temporarily or permanently fill any temporary, permanent or new position within the bargaining unit.

This agreement was ratified on the 16th of October 2018, on behalf of the Parties.

FOR THE UNION

Steve Holland

President, C.U.P.E. Local 101

FOR THE CORPORATION

William C. Coxhead

MEMORANDUM OF AGREEMENT

BETWEEN

CANADIAN UNION OF PUBLIC EMPLOYEES, LOCAL 101 (hereinafter called the Union)

AND

THE CORPORATION OF THE CITY OF LONDON (hereinafter called the Corporation)

Re: Seasonal Tourism Counsellor/Paid Lunch Under Article 10

Notwithstanding Articles 10.1(a) and 10.2(c) of the Collective Agreement the parties agree that a shift for a Seasonal Tourism Counsellor, hired between Thanksgiving and May 1st, working at the Tourism office at 267 Dundas Street will be 10:00 a.m. to 5:00 p.m. on Saturday with a paid lunch of thirty (30) minutes.

Except as outlined above, all terms and conditions of the Collective Agreement continue to apply.

This agreement was ratified on the 16th of October 2018, on behalf of the Parties.

FOR THE UNION

Steve/Holland

President, CUPE Local 101

FOR THE CORPORATION

William C. Coxhead

BETWEEN

THE CORPORATION OF THE CITY OF LONDON (the "Corporation")

AND

THE CANADIAN UNION OF PUBLIC EMPLOYEES AND ITS LOCAL 101 (the "Union")

Re: Referral to Arbitration of a Job Evaluation Matter

Notwithstanding Article 16.2 of the Collective Agreement, in the event of a referral to arbitration of a matter in dispute by either party as permitted in the Job Evaluation Manual and Terms of Reference and where the parties agree to proceed by way of a single arbitrator, the parties agree as follows:

- 1. The single arbitrator shall be appointed from a mutually agreed upon list of three (3) arbitrators who must be knowledgeable in job evaluation, currently being Robert Herman, William Kaplan and Gerry Lee. Agreed upon arbitrators shall be placed on the list in alphabetical order by surname. Appointments from the list shall be made on a rotational basis in the order of the date upon which a matter in dispute is referred to arbitration.
 - Upon ratification by both parties, the parties will contact the three arbitrators listed above to determine their willingness to participate in this agreed upon process. In the event an arbitrator on the list is unwilling or unavailable to be included in the list, the parties shall agree upon a replacement arbitrator to ensure there are three arbitrators on the agreed upon list of arbitrators at all times.
- 2. The party referring the matter in dispute to arbitration shall, within and not after, five (5) working days from the date of referral send a written invitation to the applicable arbitrator to hear the dispute. Failure to make such invitation within the time limit shall be deemed to constitute a withdrawal of the referral to arbitration.

3. The parties shall share equally in the fees and disbursements of the arbitrator.

This agreement was ratified on the 16th of October 2018, on behalf of the Parties.

FOR THE/UNION

Steve Holland

President, C.U.P.E. Local 101

FOR THE CORPORATION

William C. Coxhead

BETWEEN

THE CORPORATION OF THE CITY OF LONDON

("the Corporation")

and

LOCAL UNION NO. 101 (Canadian Union of Public Employees)

("the Union")

Re: Employees hired on or before July 22, 2015 - Job Posting and "Bumping"

The Union and the Corporation do hereby agree as follows:

- This Letter of Understanding shall be applicable only to employees with a hire date on or before July 22, 2015.
- 2. Subject to paragraph 4 below, where an employee:
 - Applies for a position outlined in a job posting in accordance with Article 8.2(a);

or

(ii) Wants to exercise the option of bumping into a position in accordance with Article 28.3(i).

The employee will be deemed to possess the necessary education, work experience and ability to perform the duties involved related to the specific position that is the subject of the job posting or bump referred to above, if the employee:

(a) Occupied a position within the classification which is the subject of the job posting or bump within and not more than three (3) years from the date of the job posting or notice of layoff;

and

(b) Worked for at least two years in the position outlined in paragraph 2(a) and has demonstrated, as determined by the Corporation, the necessary training, work experience, to successfully perform the duties of that classification.

- 3. The parties acknowledge and agree that the effect of paragraph 2 is not that the employee is automatically deemed to have the education, work experience and ability for the classification that is the subject of the job posting or bump, nor that he/she will be deemed to have the education, work experience, or ability for the purpose of future bumps or job postings for other positions within the bargaining unit.
- 4. Should the education, work experience, or ability change within the three (3) year time period noted in paragraph 2(a) above and such change is: mandated by law, professional designation or certification, paragraph 2 will not apply.
- 5. Other than as specifically provided for herein, nothing in this Letter of Understanding shall affect the Corporation's rights under Article 8.2(a).

This agreement was ratified on the 16th of October 2018, on behalf of the Parties.

FOR THE UNION

Steve Holland

President/CUPE Local 101

FOR THE CORPORATION

William C. Coxneau

APPENDIX AGREEMENT "1"

Employees Transferred From The Former Public Utilities Commission

This appendix to the Collective Agreement between the Corporation and CUPE Local 101 addresses the specific rights, benefits, contractual and statutory obligations concerning Local 101 employees who were transferred from the London Public Utilities Commission. The following provisions modify and take precedence over any conflicting references in the main Collective Agreement. For the purposes of clarity, the affected employees are listed in Schedule "A" of the Appendix.

FORMER PUC ARTICLE 13 - HOURS OF WORK

Except as otherwise provided for in this Agreement, the normal work week and the normal work day shall be as follows:

13.01 Normal Work Week

(a) Forty (40) Hour Employees:

Any five (5) consecutive days, Monday to Saturday inclusive, eight (8) hours per day, totalling forty (40) hours per week.

(b) Thirty-six and one quarter (361/4) Hour Employees:

Any five (5) days Monday to Saturday inclusive, seven and one-quarter (7¼) hours per day, totalling thirty-six and one-quarter (36¼) hours per week.

The above hours of work will be assigned in accordance with Article 10.1(b) of the main collective agreement.

13.02 Normal Work Day

(a) Forty (40) Hour Employees:

Eight (8) hours between 7:00 AM and 8:00 PM (Monday through Friday) and between 8:30 AM and 4:30 PM (Saturday), including a one-half (½) hour lunch period.

(b) Thirty-six and one quarter (361/4) Hour Employees:

Seven and one-quarter (7½) hours to be scheduled between 7:00 AM and 8:00 PM (Monday through Friday), and between 8:30 AM and 4:30 PM (Saturday) including a one-half (½) hour or one (1) hour lunch period as determined by the department manager.

The above hours of work will be assigned in accordance with Article 10.1(b) of the main collective agreement.

Occasional Shifts

An occasional shift involves employees normally working a normal work day, but who may be required at various times to work out of their normal hours to facilitate certain jobs.

Occasional shifts shall not be worked between 0800 hours and 2400 hours Saturday and 0001 hours and 2400 hours Sunday, unless the applicable overtime rate, as set out in

Article 14.01, is paid.

Occasional shifts for any individual employee shall not be scheduled for less than three (3) consecutive days excluding Saturday, Sunday and paid holidays, except that an occasional shift started the day before may continue until 0800 hours Saturday or paid holiday. All hours worked beyond 0800 hours Saturday or paid holidays shall be paid at the applicable overtime rate, as set out in Article 14.01.

If an individual employee works less than five (5) occasional shifts in any pay period, all hours worked during the normal pay period, excluding paid holidays and overtime hours, shall have the appropriate shift bonus added for the whole pay period.

An occasional shift worker returning to normal day work as per 13.02 shall not require the minimum sixteen (16) hours off between the time he or she finishes the occasional shift and returns to his or her normal day work.

Notice of required occasional shift work shall be given within the hours of the third regular shift of the employee, prior to the working of the employee's scheduled occasional shift.

Overtime - Shift Workers

Except as otherwise provided for in this Agreement, time worked in excess of the regularly scheduled shift - eight (8) hours per shift for forty (40) hour scheduled employees and seven and one-quarter (7½) hours per shift for thirty-six and one-quarter (36½) hour scheduled employees - shall be at the rate of double time.

Shift Day

All hours of a shift shall be deemed to be included in the calendar day in which the shift started

Lunch Periods

Lunch periods shall be scheduled by management but a shift employee will not be required to work more than five (5) continuous hours without a lunch break.

ARTICI F 14 - PREMIUM PAY

14.01 Overtime

Except as otherwise provided for in this Agreement, time worked in excess of the normal day, as defined in Article 13, shall be at the rate of double time.

Paid holidays shall be paid for at straight time unless otherwise agreed upon. Work done on paid holidays shall be paid at double time in addition to the regular holiday pay.

No payment shall be made for any overtime of less than one-quarter $(\frac{1}{4})$ hours straight time continuous with regular working hours, e.g. ten (10) minutes at double time equals twenty (20) minutes and no payment; fifteen (15) minutes at double time equals thirty (30) minutes and payment shall be made.

14.02 Call-out

A minimum of two (2) hours at the applicable premium rate, or the actual time worked at the applicable premium rate, whichever is greater, shall be paid for any call answered after the regular working hours except for any call answered within two (2) hours of the completion of the previous call, the time shall be considered continuous. Time shall be calculated from the time the employees leave their homes until they return for those employees designated on stand-by, trouble service, or for employees directed to the job site by the immediate supervisor. For employees directed to report to their work headquarters, time shall be calculated from the time of reporting until they return to their work headquarters or their homes as directed by their immediate supervisor.

Employees living outside the boundaries of the City shall not be paid travelling time when answering a call-out.

Any call answered in the one (1) hour preceding the one (1) hour following the employee's normal work day shall not be considered a call-out but time worked shall be paid at the applicable overtime premium rate.

14.07 Notice of Overtime

The management will endeavour to give employees required for planned overtime work, which is to be performed outside the normally scheduled hours, at least twenty (20) hours advance notice of such planned overtime during the normal work week and not later than Friday before noon for weekend overtime work.

14.08 A minimum of one (1) hour at the applicable premium rate, or the actual time worked at the applicable premium rate, whichever is greater, shall be paid for all weekend planned overtime whether worked or not unless notice of cancellation is given not later than twenty-four (24) hours previous to scheduled starting time of work.

ARTICLE 23 - SICK AND ACCIDENT PLAN

- **23.01** Sickness and non-occupational accident benefits shall be paid to permanent employees. No benefits under this Article shall be payable if the employee is disabled as a result of sickness or accident for which the employee is entitled to receive benefits under the *Workplace Safety and Insurance Act*.
- **23.02** To qualify for sick and non-occupational accident benefits, eligible employees shall notify their immediate supervisor giving a reason for their absence that is satisfactory. Employees shall call their immediate supervisor or, if their immediate supervisor is not available, their respective offices not later than fifteen (15) minutes after their scheduled starting time on the first day they are off work. If it is physically impossible for the employee to give notification within the prescribed time, he or she shall make such notification as soon as possible.
- **23.03** For periods of absence of five (5) working days or less, the employee shall provide a doctor's certificate upon return to work if requested to do so by his or her immediate supervisor during his or her absence. For all absences under this Article an "absentee slip" shall be completed by the employee.

For periods of absence of more than five (5) working days, the employee shall provide a doctor's certificate within the first fourteen (14) days of absence, and subsequently as the management may require. All such certificates shall be signed by a physician legally licensed to practice medicine and personally attending the employee for whom the certificate is submitted.

23.04 Eligible employees shall receive benefits under this Article for a period of not more than seventeen (17) weeks at full regular pay for any one disability.

If after the termination of any disability under this Article, such employee again becomes disabled due to the same or related causes, such later disability shall be treated as a continuation of the previous disability, unless such employee has completely recovered from the previous disability and was continuously and actively at work on full time and for a full pay period of at least thirty (30) days after termination of the previous disability.

- **23.05** When an employee has had four (4) incidents of sick leave in a calendar year, payment for subsequent incidents of sick leave for the balance of the year shall commence on the second consecutive working day of such leave. An incident is defined as any period of four (4) hours or more for employees in the forty (40) hour schedule and three and one-half (3 $\frac{1}{2}$) hours or more for employees in the thirty-six and one-quarter (36 $\frac{1}{4}$) hour schedule.
- **23.06** Permanent employees shall be allowed time off not to exceed four (4) hours per year for medical or dental appointments. This time may be taken as four (4) one (1) hour, two (2) two (2) hour or one (1) four (4) hour appointment(s). Time off shall be arranged with the immediate supervisor concerned prior to the appointment. Employees shall endeavour to arrange these appointments in the first or last hour(s) of the shift.

Notwithstanding the foregoing paragraph, an employee may be granted leave of absence without pay providing suitable arrangements can be made with his or her immediate supervisor.

ARTICLE 24 - MEDICAL PLANS

- **24.01** The Corporation shall pay 100% of the cost of:
 - Long Term Disability Plan, three (3) year own occupation for permanent employees.
- **24.02** (a) The following shall apply only when the employee is in receipt of L.T.D.I. benefits or *Workplace Safety and Insurance Act* benefits.
 - (b) Management shall apply for a disability waiver of premium for Life Insurance and O.M.E.R.S. pension in Article 26.
 - (c) The Corporation shall continue the payment of the employer's share of medical premiums in Articles 13.1(a), (b) and (c) and Article 24.01 as applicable for all employees in receipt of Worker's Safety and Insurance Board (WSIB) or LTD benefits.
- 24.03 (a) All of the insurance mentioned in this Article and in Article 26 shall be more particularly described in the respective policy or policies of insurance. Any dispute over payment of benefits under any such policy or policies shall be adjusted between the insured or the beneficiary under such policy and the insurer concerned, but the management shall use its best efforts to adjust and settle any such dispute.
 - (b) Every employee shall be fully responsible for keeping the Human Resources Division informed of changes in marital status or number of dependents. An employee who is entitled to a reduced hospitalization or medical benefit premium due to a change in dependency status, and who fails to notify the Human Resources Division of such change, shall be responsible for the extra premium expense paid by the Corporation on his or her behalf.

ARTICLE 25 - WORKPLACE SAFETY AND INSURANCE BENEFITS

Permanent employees off because of accidents occurring during working hours shall receive ninety percent (90%) of their regular pay from the Corporation from the first day off, provided that the employee shall receive initial treatment from a qualified medical practitioner. Any Workplace Safety and Insurance Benefits received for this period shall be paid over to the Corporation by the employee.

At the end of ninety (90) days, the employee shall be paid by the Workplace Safety and Insurance Board and Corporation payment shall cease.

ARTICLE 26 - PENSION AND INSURANCE PLANS

- 26.01 (a) The Corporation and the employees shall participate in the Ontario Municipal Employees Retirement System and Canada Pension Plan as established.
 - (b) Effective January 1, 1977 for employees who retire after January 1, 1977 and before January 1, 1981, the Corporation shall provide an O.M.E.R.S. Type I Past Service Supplementary Pension two percent (2%) formula normal retirement age of sixty-five (65) years, with the total cost paid by the Corporation.
 - (c) Effective January 1, 1981, the Corporation and the employees shall participate in the O.M.E.R.S. Type III Supplementary Pension Plan to provide for payment of a total pension commencing on retirement.
 - i) On or after his or her normal retirement date, or
 - ii) Within ten (10) years before his or her normal retirement date if the employee has completed thirty (30) years of service with the Corporation.

The Corporation shall pay the total cost of past service and the employee shall pay the total cost of future service.

The O.M.E.R.S. Type III Supplementary Benefit covers all employees who began employment with the P.U.C. prior to January 1, 1983.

(d) An employee wishing to retire under the provisions of the O.M.E.R.S. Type III Supplementary Pension Plan shall endeavour to give the immediate supervisor six (6) months notice for orderly planning to take place.

ARTICLE 27- EMPLOYMENT INSURANCE PREMIUM

27.01 The Corporation shall pay the employee's share as well as the employer's share of the cost of Employment Insurance Premiums for employees having completed six (6) months continuous service.

27.02 It is agreed that the terms of this agreement satisfy all legislative requirements related to the Employment Insurance Premium Reduction.

APPENDIX SCHEDULE "A"

Collective Agreement provisions are governed by the Appendix for the following employees:

| Paul Marshall | Steve Robinson | Ralph Thornton |
|---------------|----------------|----------------|
| Greg Traquair | Janice Glover | Lina D'Oria |
| George Negas | Bob Hannah | Debbie Melo |
| Cathy Bond | Jeff Kuchta | |

APPENDIX AGREEMENT "2" FORMER TOWN OF WESTMINSTER EMPLOYEES BENEFITS

This Appendix to the Collective Agreement between the Corporation and CUPE Local 101 addressed the specific rights, benefits and contractual obligations concerning Local 101 employees who were transferred from Former Town of Westminster. The following provisions modify and take precedence over any conflicting references in the main Collective Agreement. For the purposes of clarity, the affected employees are listed in Schedule "A" of Appendix Agreement 2.

Collective Agreement provisions are governed by the Appendix for the following employees:

Sheila Wigle Marta Semeniuk (Ponkin)

Anthony Krieger Don Simpson

BENEFITS AT A GLANCE
At December 31, 1992
CITY OF LONDON - LOCAL 101
Former Town of Westminster Employees

| Benefits | Coverage Highlights |
|-------------------------------------|--|
| WAGE LOSS REPLACEMENT Sick Leave | - 1.5 days/month cumulative, no ceiling, no cash value - if away 10 days in a month, no credit for that month |
| Long Term Disability | - following expiration of sick leave credits or 120 days, whichever is greater - 66 2/3 % of monthly earnings at date of disability , \$4,000 maximum per month - 2 years own occupation - not payable beyond age 65, retirement or termination of employment - 50 % rehabilitative income - 85 % all sources - direct offsets: CPP, WCB |
| OMERS PENSIONS | - NRA 65; Type I supplementary; Type 111 permanent partial disability |
| Compassionate Leave | - may use up to 2 days per calendar year from accumulated sick leave to attend to spouse/child/parent on special or compassionate grounds which are medically related |
| | |

| Benefits | Premium Share Arrangement |
|----------------------|---|
| Long Term Disability | 100% employer paid |
| OMERS Pension | Employer matches employee contributions which are equal to: 6% of contributory earnings up to year's Maximum Pensionable Earnings (YMPE) 7.5% above the YMPE. |

APPENDIX AGREEMENT "3"

FORMER MIDDLESEX COUNTY EMPLOYEES BENEFITS

This Appendix to the Collective Agreement between the Corporation and CUPE Local 101 addressed the specific rights, benefits and contractual obligations concerning Local 101 employees who were transferred from Former County of Middlesex. The following provisions modify and take precedence over any conflicting references in the main Collective Agreement. For the purposes of clarity, the affected employees are listed in Schedule "A" of Appendix Agreement 3.

Collective Agreement provisions are governed by the Appendix for the following employees:

Charles Brant

BENEFITS AT A GLANCE at December 31, 1992 CITY OF LONDON - Local 101

Former County of Middlesex Employees

| Long Term Disability Long Term Disability | Benefits | Coverage Highlights |
|--|-----------------------|---|
| ency | WAGE LOSS REPLACEMENT | -1 1/2 days/month cumulative |
| elifty ENCE | Sick Leave | - no sick leave credits given in any month if employee absent from duty |
| ility ENGE | | In that month: |
| ility Local Services and the services are services and the services are services and the services and the services and the services are services and the services ar | | a) for more than 15 days for any reason other than vacation, leave of |
| elity ENCE | | absence, with pay, or, |
| ENCE INCY | | b) without leave |
| ENCE incy | Long Term Disability | |
| ENCE incy | | - benefit waiting period: 17 weeks |
| ENCE ency | | - 66 2/3 % of monthly earnings (at date of disability), \$2,500 maximum |
| ENGE ency | | per month |
| ENCE incy | | - 2 years own occupation |
| ENCE incy | | - not payable beyond age 65, retirement, or termination of employment |
| ENGE sincy | | - 85 % all sources - direct offsets; CPP, WCB |
| <u>ENCE</u> incy | OMERS PENSION | - Ontario Municipal Employees Retirement System -NRA 65; Type I |
| | PAID HOLIDAYS | - 11 statutory and civic holidays as outlined in Collective Agreement - |
| | | one lieu day |
| | | - one-half working day preceding Christmas Day |
| | LEAVES OF ABSENCE | - leave without pay and without loss of seniority shall be granted to |
| | Personal/Emergency | any employee requesting such leave for good and sufficient cause |
| | | |

BENEFITS AT A GLANCE CITY OF LONDON - LOCAL 101 Former County of Middlesex At December 31, 1992

| Benefits | Coverage | Coverage Highlights |
|-----------------------|---|-------------------------------------|
| SICK LEAVE GRATUITY | - employees with more than 5 years service will receive a sick leave | s service will receive a sick leave |
| | gratuity equal to 1/2 of credits earned up to January 1, 1981 (frozen | ed up to January 1, 1981 (frozen |
| | bank only), but not more than one-half year's earnings | ıalf year's earnings |
| BENEFITS | PREMIUM SHARE | PREMIUM SHARE ARRANGEMENT |
| | EMPLOYER PAID | EMPLOYEE PAID |
| I ong Term Disability | 400% | %U |
| | 100 /0 | 0/0 |
| OMERS Pension | 6% of contributory earnings up to 6% of contributory earnings up to | 6% of contributory earnings up to |
| | year's Maximum Pensionable | year's Maximum Pensionable |
| | Earnings (YMPE) 7.5% above the | Earnings (YMPE) 7.5% above the |
| | YMPE. | YMPE. |

INDEX

| TITLE | SECTION | PAGE |
|---|-------------------------|------|
| Acting Pay Temporary Assignment | Article 9.3 | 11 |
| Anniversary Date | Article 9.5 | 12 |
| Appointments | Article 8.2 | 10 |
| Arbitration | Article 16 | 31 |
| Bargaining Committee | Article 5.1 | 4 |
| Bereavement Leave | Article 22 | 35 |
| Call-out | Article 10.9 | 16 |
| Call In & Stand- by – Remote Access | Letter of Understanding | 89 |
| Car and Clothing Allowance | Article 19 | 34 |
| Charity Chest Fund Committee | Article 5.1 | 6 |
| Classification Review Committee | Article 5.1 | 6 |
| Clothing and Safety Equipment | Schedule "B" | 64 |
| Clothing Committee | Article 5.1 | 5 |
| Convention Servicing | Letter of Understanding | 82 |
| Dental Plan | Article 13.1 | 21 |
| Disability Transfers | Article 14.7 | 29 |
| Discharge and Discipline Cases | Article 17 | 32 |
| Discrimination Prohibition | Article 4 | 4 |
| Dues | Article 2 | 2 |
| Education Allowance | Article 27 | 40 |
| Employees hired on or before July 22, 2015 – Job | | |
| Posting and "Bumping" | Letter of Understanding | 98 |
| Exclusions from the Bargaining Unit | Article 1 | 1 |
| Extended Health Care | Article 13.1 | 21 |
| Flexible Work Schedule | Letter of Understanding | 79 |
| Former Middlesex County Employees Benefits | Appendix Agreement "3" | 111 |
| Former Public Utilities Commission Employees | Appendix Agreement "1" | 100 |
| Benefits | | 400 |
| Former Town of Westminster Employees Benefits | Appendix Agreement "2" | 108 |
| General | Article 29 | 42 |
| Glasses (prescription, video display terminals) | Article 19.6 | 34 |
| Grievance Committee | Article 5.1 | 6 |
| Grievance Procedure | Article 15 | 30 |
| Group Grievance | Article 15.4 | 31 |
| Health and Safety Committee | Article 5.1 | 5 |
| Holiday Closure Periods | Letter of Understanding | 75 |
| Holidays - Paid | Article 11 | 17 |
| Hospital, Medical, Sick Leave, Pensions, Group Insurance | Article 13 | 21 |
| Hours of Work Shift, Overtime, Call Out Stand-by, Meal Allowance | Article 10 | 13 |
| Job Evaluation | Article 7 | 9 |

| 11 | | |
|---|-------------------------|----|
| Job Evaluation Committee | Article 5.1 | 5 |
| Job Evaluation – Referral to Arbitration | Letter of Understanding | 96 |
| Job Sharing | Letter of Understanding | 87 |
| Jury Duty | Article 21.1 | 35 |
| Layoff and Recall | Article 28 | 41 |
| Leaves of Absence | Article 21 | 35 |
| Leaves of Absence | Letter of Understanding | 72 |
| Life and Wage Loss Insurance Plans | Article 13.6 – 13.10 | 28 |
| Long Term Disability Details | Appendix "B" | 70 |
| Loss of Seniority | Article 6.5 | 8 |
| Management Functions | Article 3 | 3 |
| Meal Allowance | Article 10.10 | 16 |
| Municipal Law Enforcement Officer II | Letter of Understanding | 86 |
| New Positions | Article 7 | 9 |
| Ontario Works Program | Letter of Understanding | 74 |
| Overtime | Article 10.5 – 10.7 | 15 |
| Part-time Employment | Article 24 | 39 |
| Pensions-OMERS | Article 13.4 – 13.5 | 26 |
| Permanent Vacancy, Notice | Article 8.2 | 10 |
| Policy Grievance | Article 15.4 | 31 |
| Position Classification and Salary Structure | Schedule "A" | 44 |
| Probationary Employee | Article 6.2 | 7 |
| Promotion | Article 9.4 | 11 |
| Protection of Employee's Position | Article 18 | 33 |
| Reclassification and Revision of Positions and New Positions | Article 7 | 9 |
| Recognition and Scope | Article 1 | 1 |
| Rehabilitative Assignments and Workplace Safety and Insurance | Article 14 | 28 |
| Remuneration and Interview | Article 9 | 11 |
| Retained Seniority | Article 6.6 | 8 |
| Retiree's Benefits | Article 13.1(e) | 22 |
| Retirees Benefits | Appendix "A" | 69 |
| Return to Work Committee | Article 5.1(j) | 6 |
| Rights, Benefits and Privileges Letter | | 68 |
| Seniority | Article 6 | 7 |
| Seniority List | Article 6.3 | 7 |
| Shift Premiums | Article 10.4 | 15 |
| Shifts | Article 10.2 | 14 |
| Sick Leave | Article 13.2 | 23 |
| Staff Changes, Additions, and Promotions | Article 8 | 9 |
| , , | | |

| Stand-by | Article 10.12 | 16 |
|---|-------------------------|----|
| Strikes or Lockouts | Article 26 | 40 |
| Student Employment | Article 25 | 39 |
| Temporary Assignment | Article 23 | 36 |
| Testing | Letter of Understanding | 77 |
| Terms and Definitions | Article 30 | 42 |
| Term of Agreement | Article 31 | 43 |
| Terms of Dedicated Presidency for Local 101 | Letter of Understanding | 84 |
| Trial Period | Article 8.3 | 10 |
| Union Conference, Convention or Seminar | Article 20 | 34 |
| Union - Management Committee (The) | Article 5.1 | 4 |
| Union - Management Responsibilities | Article 4 | 4 |
| Union Representation | Article 5 | 4 |
| Union Security and Check-off | Article 2 | 2 |
| Vacations | Article 12 | 18 |
| Vision care | Article 13.1(c) | 22 |
| Workplace Safety and Insurance Benefits | Article 14.5 | 29 |